Columbia Law School Scholarship Archive

Faculty Scholarship

Faculty Publications

2017

Criminal Deterrence: A Review of the Literature

Aaron Chalfin University of Pennsylvania

Justin McCrary Columbia Law School, jmccrary@law.columbia.edu

Follow this and additional works at: https://scholarship.law.columbia.edu/faculty_scholarship

Part of the Criminal Law Commons, and the Law and Economics Commons

Recommended Citation

Aaron Chalfin & Justin McCrary, *Criminal Deterrence: A Review of the Literature*, 55 J. ECON. LIT. 5 (2017). Available at: https://scholarship.law.columbia.edu/faculty_scholarship/3203

This Article is brought to you for free and open access by the Faculty Publications at Scholarship Archive. It has been accepted for inclusion in Faculty Scholarship by an authorized administrator of Scholarship Archive. For more information, please contact scholarshiparchive@law.columbia.edu, rwitt@law.columbia.edu.

Criminal Deterrence: A Review of the Literature[†]

AARON CHALFIN AND JUSTIN MCCRARY*

We review economics research regarding the effect of police, punishments, and work on crime, with a particular focus on papers from the last twenty years. Evidence in favor of deterrence effects is mixed. While there is considerable evidence that crime is responsive to police and to the existence of attractive legitimate labor-market opportunities, there is far less evidence that crime responds to the severity of criminal sanctions. We discuss fruitful directions for future work and implications for public policy. (JEL J64, K42)

1. Introduction

 $T^{\rm he}$ day-to-day work of individuals employed in law enforcement, corrections, and other parts of the criminal-justice system involves identifying, capturing, prosecuting, sentencing, and incarcerating offenders. Perhaps the central function of these activities, however, is deterring individuals from participating in illegal activity in the first place. Deterrence is important not only because it results in lower crime but also because, relative to incapacitation, it is cheap. Offenders who are deterred from committing crime in the first place do not have to be identified, captured, prosecuted, sentenced, or incarcerated. For this reason, assessing the degree to which potential offenders are deterred by either carrots

(better employment opportunities) or sticks (more intensive policing or harsher sanctions) is a first-order policy issue.

The standard economic model of criminal behavior draws on a simple expected utility model introduced in a seminal contribution by the late Gary Becker. This model envisions crime as a gamble undertaken by a rational individual. According to this framework, the aggregate supply of offenses will depend on social investments in police and prisons as well as on labor-market opportunities that increase the relative cost of time spent in illegal activities.

Using Becker's work as a guide, a large empirical literature has developed to test the degree to which potential offenders are deterred. The papers in this literature fall into three general categories. First, a number of papers consider the responsiveness of crime to the probability that an individual is apprehended. This concept has typically been operationalized as the study of the sensitivity of crime to police, in particular

^{*} Chalfin: University of Pennsylvania. McCrary: University of California, Berkeley and NBER.

[†]Go to https://doi.org/10.1257/jel.20141147 to visit the article page and view author disclosure statement(s).

police manpower or policing intensity. A second group of papers studies the sensitivity of crime to changes in the severity of criminal sanctions. This literature assesses the responsiveness of crime to sentence enhancements, strikes laws, capital-punishment three regimes, and policy-induced discontinuities in the severity of sanctions faced by particular individuals. The third group of papers examines the responsiveness of crime to local labor-market conditions, generationally operationalized using either the unemployment rate or a relevant market wage. This literature seeks to determine whether crime can be deterred through the use of positive incentives rather than punishments.

The papers in each of these literatures can be viewed as measuring the degree to which individuals can be deterred from participation in criminal activity. Each of the literatures is vast and it is not unreasonable to suggest that each could merit a separate review. A challenge remains to characterize the pattern of the empirical findings and explain why individuals appear to be more responsive (and thus more deterrable) along certain margins than along others. In this article, we provide a brief review of each of the three literatures introduced above with the intention of rationalizing several apparently divergent findings.

We are not the first to review the deterrence literature. Indeed, in last decade we count a number of comprehensive reviews on the subject including, but not limited to, Levitt and Miles (2006), Tonry (2008), Durlauf and Nagin (2011), Nagin (2013), and Chalfin and Tahamont (forthcoming). We attempt to differentiate our review in several ways. First, we have tried to synthesize research carried out by economists, as well as criminologists. In this goal, we are not alone. However, we are hopeful that by highlighting in the *JEL* research from criminology that is typically unknown to economists, we will help to further integrate the two disciplines. Second, interest in deterrence research has multiplied rapidly over the last few years, with a number of important studies having been published in the last year or two alone. Accordingly, we have done our best to include references to the newest and most cutting-edge research. Finally, in this review, we cover a topic that is often omitted from reviews of the deterrence literature—the role of labor markets in deterring crime via "carrots" rather than "sticks." The remainder of the paper is laid out as follows: section 2 considers research on the effect of police on crime, section 3 considers the effect of prison and/or sanctions on crime, and Section 4 considers the responsiveness of crime to local labor-market conditions. Section 5 concludes.

2. Theories of Deterrence

Deterrence is an old idea and has been discussed in academic writing at least as far back as eighteenth-century treatises by Adam Smith (1776), Jeremy Bentham (1789), and Cesare Beccaria (1764). There are three core concepts embedded in theories of deterrence-that individuals respond to changes in the certainty, severity, and celerity (or immediacy) of punishment. Interestingly, in the criminological tradition, deterrence is often characterized as being either general or specific, with general deterrence referring to the idea that individuals respond to the *threat* of punishment and specific deterrence referring to the idea that individuals are responsive to the actual *experience* of punishment. Economics prefers different terminology, reserving the term deterrence for what the criminologist calls general deterrence and describing specific deterrence as a change in information or, perhaps more exotically, a change in preferences themselves. In this section, we briefly characterize the way economists have formalized these concepts. In general, economic theories of deterrence have focused more

heavily on certainty and severity. However, recent writing has increasingly characterized deterrence as part of a dynamic framework in which offender behavior is sensitive to their time preferences (Polinsky and Shavell 1999 and Lee and McCrary forthcoming).

2.1 Economic Models of Crime

The earliest formal model of criminal offending in economics can be found in Becker's seminal 1968 paper, "Crime and Punishment: An Economic Approach." The crux of Becker's model is the idea that a rational offender faces a gamble. He can either choose to commit a crime and thus receive a criminal benefit (albeit with an associated risk of apprehension and subsequent punishment) or not to commit a crime (which yields no criminal benefit but is risk free). The expected cost of committing a crime is a function of the offender's probability of apprehension, p, and the severity of the sanction that he will face upon apprehension, f. To be more specific, the individual can be said to face three potential outcomes, each of which delivers a different level of utility: (1) the utility associated with the choice to abstain from crime, U_{nc} ; (2) the utility associated with choosing to commit a crime that does not result in an apprehension, U_{c1} ; and (3) the utility associated with choosing to commit a crime that results in apprehension and punishment, U_{c2} . In such a formulation, the individual chooses to commit a crime if, and only if, the following condition holds:

(1)
$$(1-p)U_{c1} + pU_{c2} > U_{nc}$$

That is, crime is worthwhile so long as its expected utility exceeds the utility from abstention.¹

In addition to the clear role played in this model by the probability of apprehension, p, the formulation also suggests the importance of two additional exogenous factors that could influence U_{c2} and U_{nc} . Crime becomes more attractive when the disutility of apprehension is slight (e.g., less unpleasant prison conditions) and it becomes less attractive when the utility of work is high (e.g., a low unemployment rate or a high wage). Becker operationalizes the disutility associated with capture using a single exogenous variable, f, which he refers to as the severity of the criminal sanction upon capture. Typically, fis assumed to refer to something like a fine, the probability of conviction, or the length of a prison sentence.² To a large degree, then, government maintains control over U_{c2} .

The utility associated with abstaining from crime, U_{nc} , is principally a function of the individual's ability to derive utility from non-illicit activities. In practice, this is typically thought of as the wage that can be earned in the legal labor market. When the legal wage rises, U_{nc} rises, thus reducing the relative benefit of criminal activity. It is fair to say that while government maintains some control over U_{nc} , it does so to a lesser extent than it does over the utility of punishment, U_{c2} .

Using these ideas, Becker rewrites (see footnote 16) the expected utility confronting an individual contemplating crime as

(2)
$$EU = pU(Y-f) + (1-p)U(Y)$$

where Y represents the income associated with getting away with crime.³ In this

³As Becker is careful to say, income "monetary and psychic."

¹The "if and only if" holds if we maintain that the case of $(1-p)U_{c1} + pU_{c2} = U_{nc}$ implies no crime, an unimportant assumption we make henceforth to simplify discussion.

 $^{^{2}}$ In principle, *f* can be a function of many different characteristics of the sanction including the length of the sentence, the conditions under which the sentence will be served, and the degree of social stigma that is attached to a term of incarceration, all of which are likely heterogeneous among the population.

formulation, crime occurs if and only if $EU > U_{nc}$. Equivalently, we can define an indifference point, Y^* , such that crime occurs if and only if $Y > Y^*$. It is easy to see that

(3)
$$\frac{U(Y^*) - U_{nc}}{|U(Y^* - f) - U_{nc}|} = \frac{p}{1 - p}.$$

Several important ideas are embedded in (3). First, for the individual to elect to engage in crime, the gain relative to its loss must exceed the odds of capture. Dividing the numerator and denominator of the left side by U_{nc} yields a natural interpretation, in terms of percentages. Consider a criminal opportunity where capture is n times as likely as not. Crime occurs if the anticipated percent improvement in utility associated with getting away with it is more than n times as large as the anticipated percent reduction in utility associated with apprehension. Second, an increase in p unambiguously reduces the likelihood of crime, as this increases the right-hand side of (3). Third, an increase in f unambiguously reduces the likelihood of crime as long as $U'(\cdot) > 0$, as this decreases the left-hand side of (3).

Under risk neutrality, the equation (3) simplifies. Define *a* as the income associated with abstaining from crime, i.e., $U(a) = U_{nc}$; define c = f - b > 0 as the effective cost of punishment. Further define income, Y = a + b and $Y^* = a + b^*$, where *b* is the criminal benefit and b^* is the criminal benefit at which the individual is indifferent between crime and abstention. Then equation (3) reduces to

$$b^* = c \frac{p}{1-p}$$

This simplified version of the Becker model is the starting point of the dynamic analysis in Lee and McCrary (forthcoming).

A somewhat different focus can be found in Ehrlich (1973), where the notion of the opportunity cost of engaging in crime is front and center. Perhaps unsurprisingly, labor economists have found it particularly attractive to view crime as a time-allocation choice, and this type of formulation is found in several prominent papers including Lemieux, Fortin, and Frechette (1994); Grogger (1998); Williams and Sickles (2002); and Burdett, Lagos, and Wright (2004), among others.⁴

The typical time-allocation model of crime considers a consumer facing a constant market wage and diminishing marginal returns to participation in crime. This consumer maximizes a utility function that increases in both leisure (L) and consumption (C), where consumption is financed by time spent engaged in either legitimate employment (h_m) at a market wage (w) or time spent in crime (h_c) with a net hourly payoff of r. The consumer's constrained optimization problem is to maximize his utility function, U(C, L), subject to the consumption and time constraints:

$$(4) C = w h_m + r h_c + I$$

$$(5) L = T - h_m - h_c.$$

In (4), consumption is shown to be equal to an offender's legitimate income plus his non-legitimate income.⁵ In (5), T is the individual's time endowment and leisure is the remaining time after market work and time spent in crime is accounted for.⁶ The parameter r reflects the criminal benefit but it

⁴For further details, see Gronau (1980).

⁶Grogger assumes that the returns to crime diminish as the amount of time devoted to criminal activity increases—i.e., there is a concave function $r(\cdot)$ that translates hours spent participating in crime into income. Diminishing returns implies that those engaging in criminal activity first commit crimes with the highest expected payoffs (lowest probability of getting caught and highest stakes) before exploring less lucrative opportunities. However, this need not be true.

 $^{^{5}}I$ represents nonlabor income.

also reflects the costs of committing crime, namely the risk of capture and the expected criminal sanction if captured. In other words, r can be thought of as the wage rate of crime, net of the expected costs associated with the criminal-justice system. In this way, criminal sanctions drive a wedge between the consumer's productivity in offending and his market wage, in turn, incentivizing market work over crime.⁷

An interesting question in both Becker's model and Ehrlich's model is whether individuals are more deterred by increases in por f. Becker addresses this in a straightforward way by asking whether the expected utility of crime is decreased more by a small percent increase in p or an equivalent percent increase in f. This makes sense because participation in crime should be monotonic in its expected utility. Becker's analysis shows that p is more effective if and only if $U''(\cdot) > 0$, i.e., if and only if individuals were risk preferring.⁸ If individuals are averse to risk, increasing f is more effective than increasing p, and if individuals are risk neutral, then f and p are equally effective. Becker notes (footnote 12) that this conclusion is the opposite of that given by Beccaria regarding the effectiveness of punishment versus capture, and that the conclusion is similarly at odds with contemporaneous views of judges.

⁷In order for an individual to commit any crime at all, there are two necessary and sufficient conditions. First, the marginal return to the first instant of time supplied to crime must exceed the individual's valuation of time (in terms of how much consumption the person would be willing to forgo for more time) when all time is devoted to nonmarket, noncrime activities. Second, the marginal return to crime for the first crime committed must exceed the individual's market wage. Thus, those who can command high wages or those who place very high value on time devoted to nonmarket/noncriminal uses will be the least likely to engage in criminal activity.

⁸Note, however, the observant criticism of Brown and Reynolds (1973), showing that this clean conclusion is the result of the modeling assumption that the baseline utility is that of getting away with crime. The model of Lee and McCrary (forthcoming) emphasizes the dependence of this conclusion on the time preferences of the individual.⁹ Intuitively, it seems like it would be hard to deter an impatient individual using a prison sentence, since most of the disutility of a prison sentence is borne in the future. Lee and McCrary propose modeling crime using a modification of the basic job search model in discrete time with an infinite horizon. Risk-neutrality is assumed, yet individuals in this model have very different responses to the capture and punishment.

In their model, criminal opportunities are independent draws from an identical "criminal benefit" distribution with distribution function F(b). The individual learns of an opportunity each period and must decide whether to take advantage of it. If the individual engages in crime and is caught, she is imprisoned for S periods, where S is an independent draw from an identical sentence-length distribution. As in the Becker model, capture occurs with probability p. If the individual abstains from crime, she obtains flow utility a and faces the same problem the next period. If she commits crime and is not caught, she obtains flow utility a + b and faces the same problem next period. Finally, if she commits crime and is caught, she obtains flow utility a - c for S periods, before confronting the same problem at the conclusion of her sentence.

The criminal benefit at which the individual is indifferent between crime and abstention is given by

(6)
$$b^* = c \frac{p}{1-p}$$

+ $\nu \Big\{ c \frac{p}{1-p} + p \int_{b^*}^{\infty} (1 - F(z)) dz \Big\},$

⁹Further details regarding the Lee and McCrary model are given in McCrary (2010).

where $\nu = E\left[\sum_{s=1}^{S-1} \delta^s\right]$ is a summary parameter governing how the distribution of sentences affects decision making and δ is the discount factor.^{10, 11}

As in Becker's static model, crime is reduced by increases in p and increases in c. The added feature of this model, however, is that crime is also reduced by increases in sentence lengths, and this behavioral mechanism is modulated by time preferences. As is intuitive, patient individuals are quite responsive to increases to sentence lengths, but impatient individuals show much more muted responses. In the limit as the discount factor approaches zero, the individual is arbitrarily more responsive to capture than to punishment. The Lee and McCrary model thus provides a simple way to reintroduce older ideas regarding the importance of celerity into a Becker model.¹²

Ultimately, the models proposed by Becker and Ehrlich yield three main behavioral predictions: (1) the supply of offenses will fall as the probability of apprehension rises, (2) the supply of offenses will fall as the severity of the criminal sanction increases, and (3) the supply of offenses will fall as the opportunity cost of crime rises. In other words, behavioral changes can be brought about either using carrots (better employment opportunities) or sticks (criminal-justice inputs). The following section connects these core predictions to the empirical literatures that have

¹⁰ For example, if we take *S* to be geometric, i.e., *S* has support 1, 2, ..., ∞ and $P(S = s) = q(1 - q)^{s-1}$, where *q* is the per period release probability, then standard results using infinite series show that $\nu = (1 - q)\delta/(1 - (1 - q)\delta)$. Interestingly, this shows that under a geometric distribution for sentence lengths, reducing the probability of release is equivalent to increasing the individual's patience.

¹¹Equation (6) is not an explicit equation for b^* , but it can be viewed as defining an implicit function. We can use numerical methods to solve for b^* (e.g., Newton's method works well), and comparative statics are straightforward using the implicit function theorem.

¹²For related modeling ideas from criminology, see Nagin and Pogarsky (2001), for example.

sought to test whether these predictions hold in the real world.

2.2 Perceptions and Deterrence

Because economic models of offending are microeconomic models that make predictions about individual behavior, our discussion of deterrence would be incomplete without a discussion of how individuals perceive risks and, especially, whether risk perceptions mirror reality. Given the scope of this review, our discussion of perceptions is necessarily brief. We direct the reader to an excellent review of this literature by Apel (2013) for a more detailed accounting of the perceptual-deterrence literature.

Perceptual deterrence is important because the vast majority of the empirical-deterrence literature operationalizes Becker's model of crime by studying the responsiveness of crime to particular policy variables, such as the number or productivity of police or the punitiveness of sanctions. This approach was initially borne out of the inadequacy of data needed to test the microfoundations of the Becker model, but has the advantage of having generated a dense literature that is practical and policy-relevant. Given that the literature studies the effect of policy variables, an important intermediate outcome and indeed a precursor to identifying deterrence is the extent to which potential offenders are aware that policy has changed (Waldo and Chiricos 1972, Nagin 1998, and Apel 2013). Apel (2013) characterizes the link between actual and perceived deterrence as involving a series of considerations that include both threat communication, the degree to which a change in the certainty or the severity of a sanction is communicated or advertised, and risk perceptions, the individual's perceived risk of being apprehended and punished. Crucially, risk perception is not assumed to be stable and indeed an important literature has arisen that seeks to understand how

offenders update risk perceptions in response to experience (see Apel and Nagin 2011 for a comprehensive review on the subject).

Ultimately, one of the most important questions for perceptual-deterrence research is the degree of correspondence between actual and perceived risks. If perceptions closely mirror reality, then using policy shocks to learn about the magnitude of deterrence is straightforward. However, to the extent that changes in policy often go unnoticed by potential offenders, the outcomes of policy research will tend to be of limited value in studying deterrence. Consider, for example, a policy that increases the number of undercover police officers who are assigned to patrol a city's transit system. Assuming that policy is unannounced and, even if it is announced, that the news does not easily trickle down to potential offenders, it is difficult to imagine how deterrence will accrue. It may well be that the policy begins to be noticed by offenders as they hear about cases in which undercover officers have made arrests or if they have an acquaintance who has been arrested in this way. However, it seems likely that such information will generate deterrence only via a substantial temporal lag. Indeed, it seems likely that a highly visible change in the number of uniformed officers or, alternatively, a well-advertised policy to increase the number of undercover officers, will generate a greater deterrence effect, even if the actual intervention is no different.

The recent literature that links actual and perceived risks is relatively small. Important recent work includes that of Kleck et al. (2005) and Kleck and Barnes (2013), who conducted a telephone survey of 1,500 adults in fifty-four large urban counties in the United States. They asked each individual to estimate case clearance rates, the probability of serving time in prison, and maximum sentence for several different serious felonies. Comparing perceived risks to actual risks, they found little evidence of any correlations, a finding that extends to police manpower as well. Research by Lochner (2007) using the National Longitudinal Survey of Youth (NLSY) comes to a qualitatively similar conclusion reporting evidence of a significant, albeit weak, relationship between actual and perceived risks of apprehension. Likewise, in an application to drug use, a very common crime resulting in arrest, MacCoun et al. (2009) report that individuals living in states that have decriminalized marijuana often do not have any awareness of this and continue to believe that they can be jailed for marijuana possession. These studies are characterized by Apel (2013) as being discouraging for deterrence research. However, as each of the studies surveyed the general population, most of whom are uninvolved in crime, such research may have poor external validity. The best evidence on perceptions among a sample of active offenders comes from Lochner (2007), who reports that NLSY youth who self-report criminal involvement do, on average, have more accurate perceptions about arrest risks than noncriminally involved youth.

A second strain of research considers whether offenders change their risk perceptions in response to a past arrest. One flavor of this research has compared risk perceptions among individuals who reported more frequent arrest conditional upon offending (i.e., less successful offenders) to individuals who reported fewer arrests per offense (i.e., more successful offenders). This literature tends to find robust evidence of an association between more frequent arrest and a higher perceived probability of capture (Paternoster and Piquero 1995, Piquero and Pogarsky 2002, Pogarsky and Piquero 2003, and Carmichael and Piquero 2006). A parallel literature has found that risk perceptions are also informed by the experience of acquaintances (Piquero and Pogarsky 2002). Unfortunately, there are a number of conceptual issues that make this

literature difficult to interpret. Most notably, since these associations arise from cross-sectional data, it is not possible to discern cause from correlation. In particular, it is plausible that more successful offenders have lower perceived arrest probabilities for reasons that are a function of personality and largely unrelated to experience. In response to this concern, a more recent literature uses panel data to measure "updating"—the idea that individuals change their prior risk perceptions on the basis of whether or not they are apprehended in an earlier period. This literature has also tended to find robust evidence that risk perceptions are sensitive to actual experience (Pogarsky, Piquero, and Paternoster 2004; Pogarsky, Kim, and Paternoster 2005; Matsueda, Kreager, and Huizinga 2006; and Anwar and Loughran 2011). Several more specific findings from this literature are worth noting. First, while perceptions are responsive to experience, offending is not always responsive to perceptions, implying that at least a portion of offending may be idiosyncratic and perhaps undeterrable in a stable policy regime. Second, less experienced offenders are especially sensitive to the experiences of peers, which is sensible as they may not have sufficient history upon which to draw conclusions. Third, there is evidence that the general public, along with less frequent offenders, tend to overestimate their arrest risk and adjust their risk perceptions downward as they offend and recognize that the risk of apprehension is lower than they previously believed. An important corollary to this is that risk perceptions are more sensitive to experience early in one's criminal career, with the deterrence value of an arrest declining with experience (Anwar and Loughran 2011).

Broadly speaking, the perceptual-deterrence literature provides several reasons to be optimistic that meaningful deterrence effects can exist and can be particularly salient among younger offenders who have yet to commit to a criminal career. The best available evidence suggests that the experience of arrest does lead to an increase in the perceived likelihood of being apprehended for a future crime. What is less clear is whether perceived risks change in response to policy inputs that have more diffuse impacts and whether advertising sanctions can be a sufficiently credible threat—a proposition we discuss in further detail in the subsequent empirical section of this paper.

2.3 Deterrence versus Incapacitation

Generally speaking, there are two mechanisms through which criminal-justice policy reduces crime: deterrence and incapacitation. When by virtue of a policy change individuals elect not to engage in crime they otherwise would have in the absence of the change, we speak of the policy deterring crime. On the other hand, a policy change may also take offenders out of circulation as, for example, with pretrial detention or incarceration, preventing crime by incapacitating individuals. The incapacitation effect can be thought of as the mechanical response of crime to changes in criminal-justice inputs. While deterrence can arise in response to any policy that changes the costs or benefits of offending, incapacitation arises only when the probability of capture or the expected length of detention increases.

The existence of incapacitation effects has profound implications for the study of deterrence. In particular, while research that considers the effect of a change in the probability of capture will, generally speaking, identify a mixture of deterrence and incapacitation effects, research that considers changes in the opportunity cost of crime is more likely to isolate deterrence. Likewise, while research on the effect of sanctions typically results in a treatment effect that is a function of both deterrence and incapacitation, clever research designs have been used to identify the effect of an increase in the severity of a sanction that is unlikely to result in an immediate increase in incapacitation.

For each literature discussed in this paper, we provide a discussion of the degree to which empirical estimates can be interpreted as providing evidence of deterrence as distinct from incapacitation and, in some cases, other behavioral effects. However, it is important to note that deterrence is itself a black box. In order to empirically observe a behavioral response of crime to a particular policy level, it must be the case that potential offenders perceive that the cost of committing a crime has changed (Nagin 1998, Durlauf and Nagin 2011, and Nagin 2013). Moreover, the behavioral response of crime will depend on the accuracy of those perceptions. To wit, an intervention that successfully convinces potential offenders that the expected cost of crime has increased, regardless of whether this is actually the case, will likely reduce crime. The challenge for cost-effective public policy is to optimally trade off between police and prisons so as to maximize perceptual and, as such, actual deterrence.

3. Police and Crime

Becker's prediction that the aggregate supply of crime will be sensitive to society's investment in police arises from the idea that an increase in police presence, whether it is operationalized through increased manpower or increased productivity, raises the probability that an individual is apprehended for having committed a particular offense. To the extent that potential offenders are able to observe an increase in police resources and perceive a correspondingly higher risk to criminal participation, crime is expected to decline through the deterrence channel.

Empirically, the challenge for this literature is that changes in the intensity of policing are generally not random. As a result, it is difficult to identify a causal effect of police on crime using natural variation in policing. An additional, more conceptual issue is that the responsiveness of crime to police may also reflect an important role for incapacitation. This arises from the idea that police tend to reduce crime mechanically, even in the absence of a behavioral response, by arresting offenders who are subsequently incarcerated and incapacitated.13 The extent to which investments in police are cost effective depends, in large part, on the degree to which police deter rather than simply incapacitate offenders. In this section, we consider the responsiveness of crime to both police manpower and police tactics, broadly defined. For each literature, we discuss the challenges with respect to both econometric identification as well as interpretation of the resulting parameters as evidence in favor of deterrence.

3.1 Police Manpower

A large literature has used city- or statelevel panel data and, recently, a variety of quasi-experimental designs to estimate the elasticity of crime with respect to police manpower.¹⁴ This literature is ably summarized by Cameron (1988), Nagin (1998), Eck and Maguire (2000), Skogan and Frydl (2004), and Levitt and Miles (2006), all of whom provide extensive references.

The early panel-data literature tended to report small elasticity estimates that were rarely distinguishable from zero and sometimes even positive, suggesting perversely that police increase crime.¹⁵ The ensuing

¹⁵Papers in this literature employ a wide variety of econometric approaches. Early empirical papers such as Ehrlich (1973) and Wilson and Boland (1978) focused on the cross-sectional association between police and crime.

¹³In this context, deterrence can arise either from a general decrease in offending or from a shift towards less productive but correspondingly less risky modes of offending—for example, a shift from robbery to larceny.

¹⁴This elasticity can be thought of as a reduced-form parameter that captures both deterrence effects as suggested by neoclassical economic theory, as well as incapacitation effects that arise when offenders are incarcerated and thus constrained in their ability to offend.

discussion in the literature was whether police reduce crime at all. Beginning with Levitt (1997), an emerging quasi-experimental literature has argued that simultaneity bias is the culprit for the small elasticities in the paneldata literature.¹⁶ The specific concern articulated is that if police are hired in anticipation of an upswing in crime, then there will be a positive bias associated with regression-based strategies, masking a true negative elasticity. The recent literature has therefore generally focused instead on instrumental variables (IV) strategies designed to overcome this bias.

The first plausible instrumental variable to study the effect of police manpower on crime was proposed by Levitt (1997). Leveraging data on the timing of mayoral and gubernatorial elections, Levitt provides evidence that in the year prior to a municipal or state election, police manpower tends to increase, presumably due to the desire of elected officials to appear to be "tough on crime." The exclusion restriction is that, but for increases in police manpower, crime does not vary cyclically with respect to the election cycle. Using data from fifty-seven cities spanning 1972–97, Levitt reports very small least-squares estimates of the effect of police and crime that are consistent with the prior literature. However, IV estimates are large and economically important, with elasticities ranging from moderate in magnitude for property crimes (-0.55 for burglary)and -0.44 for motor vehicle theft) to large in magnitude for violent crimes such as robbery (-1.3) and murder (-3). Ultimately, following a reanalysis of the data by McCrary (2002), the IV coefficients reported by Levitt were found to be insignificant after a problem with weighting was addressed. The insignificance of the coefficients is ultimately driven by the fact that the first-stage relationship between election cycles and police hiring is weak, complicating both estimation and inference.

Levitt (1997) has given rise to a series of related papers that seek to identify a national effect of police manpower on crime by isolating conditionally exogenous within-city variation in police staffing levels. These papers include Levitt (2002), which uses variation in firefighter numbers as an instrument for police manpower; Evans and Owens (2007), who instrument for police manpower using the size of federal Community Oriented Policing Services (COPS) grants awarded to cities to promote police hiring; and Lin (2009), who instruments for changes in police manpower using the idea that US states have differential exposure to exchange-rate shocks depending on the export intensity of local industry. These strategies consistently demonstrate that police do reduce crime.¹⁷ However, the estimated elasticities display a wide range, roughly -0.1 to -2, depending on the study and the type of crime. Moreover, relatively few of the estimated elasticities are significant at conventional levels of confidence, reflecting a great deal of sampling variability and the use of relatively weak instruments. In many cases, extremely large elasticities (i.e., those larger than one in magnitude) cannot be differentiated from zero. Overall, Chalfin and McCrary (forthcoming) characterize the pattern of the cross-crime elasticities as, in general, favoring a larger effect of police on violent crimes than on property crimes, with especially large effects of police on murder, robbery, and motor vehicle theft.¹⁸

¹⁶Some of the leading examples of quasi-experimental papers are Levitt (2002), Di Tella and Schargrodsky (2004), Klick and Tabarrok (2005), Evans and Owens (2007), Lin (2009), and Machin and Marie (2011).

¹⁷Notably, Worrall and Kovandzic (2007) report no reduced-form relationship between COPS grants and crime. However, their analysis is based on a smaller sample of cities than the analysis of Evans and Owens (2007).

¹⁸This pattern is found in several prominent panel data papers, in particular Levitt (1997), Evans and Owens (2007), and Chalfin and McCrary (forthcoming), each of which report especially large elasticity estimates for murder (-0.6 to -0.8) and robbery (-0.5 to -1.4).

A second noteworthy contribution to the modern police manpower literature is that of Marvell and Moody (1996), who leverage the concept of Granger causality to explore the extent to which police manpower is, in fact, responsive to changes in crime. The motivation behind such an approach is that if crime is responsive to lagged police but police staffing is not responsive to lagged crime, then the case for instrumental variables is weakened considerably. Finding no evidence of a link between lagged crime rates and current police staffing levels at either the state or city level, Marvell and Moody estimate the responsiveness of crime to police using a standard two-way fixed-effects model and report elasticities that are fairly small in magnitude (ranging from -0.15 for burglary to -0.30 for motor vehicle theft) and are more consistent with the early least-squares literature than the IV literature that has proliferated in recent years.

Ultimately, the Granger causality exercise is subject to the same omitted variables bias concerns that plague any least squares regression model, and is therefore of dubious value in establishing causality. Nevertheless, the weak evidence of a link between lagged crime and current police staffing presented in Marvell and Moody is, in our view, underappreciated. Given the large discrepancy between Marvell and Moody's estimates and those in Levitt (1997), which use the same underlying data, one of two propositions must be true: (1) Marvell and Moody's estimates of the effect of lagged crime on police manpower are biased due to the exclusion of important omitted variables, or (2) There is no simultaneity bias between police and crime discrepancies between least squares and IV estimates are instead driven by measurement errors in either police staffing or measures of UCR index crimes. This is an idea that is dealt with in detail in Chalfin and McCrary (forthcoming). Leveraging two potentially independent measures of police manpower (one

from the FBI's Uniform Crime Reports and another from the US Census's Annual Survey of Government Employment) for a sample of 242 US cities over a fifty-one-year time period, Chalfin and McCrary construct measurement error corrected IV models using one measure of police as an instrument for the other. Their principal finding is that elasticities reported in the recent IV literature can be replicated by simply correcting for measurement errors in police data and without explicitly addressing the possibility of simultaneity bias. The resulting implication is that Marvell and Moody's basic inference regarding the lack of causality running from crime to police manpower may be correct. A related contribution in Chalfin and McCrary is to estimate police elasticities with remarkable precision, reporting elasticities of -0.67 ± 0.48 for murder, -0.56 ± 0.24 for robbery, -0.34 ± 0.20 for motor vehicle theft, and -0.23 ± 0.18 for burglary.

While the majority of the police manpower literature uses aggregate data, there is a corresponding literature that assesses the impact of police on crime using natural experiments in a particular jurisdiction. An early account of such a natural experiment is found in Andenaes (1974), who documents a large increase in crime in Nazi-occupied Denmark after German soldiers dissolved the entire Danish police force (Durlauf and Nagin 2011 and Nagin 2013). Modern literature has found similarly large effects. In particular, DeAngelo and Hansen (2014) document an increase in traffic fatalities that occurred in the aftermath of a budget cut in Oregon that resulted in a mass layoff of state troopers. Similarly, Shi (2009) reports an increase in crime in Cincinnati, OH, in the aftermath of an incident in which police used deadly force against an unarmed African American teenager.¹⁹

¹⁹As Shi (2009) notes, the police response to the riot was to reduce productivity disproportionately in riot-affected neighborhoods.

3.2 Police Deployment and Tactics

The police manpower literature is informative with respect to the aggregate response of crime to increases in police staffing. However, the aggregate manpower literature leaves many interesting and important questions unanswered. In particular, to what extent do the estimated elasticities reflect deterrence? Likewise, what is the specific mechanism that leads to deterrence? If the mechanism is based on perceptual deterrence-the idea that offenders observe an increase in police presence and adjust their behavior accordingly—then it should be the case that offending is especially sensitive to large and easily observed changes in police deployment and tactics. To address these questions, a related literature that is found mostly in criminology has studied the effect of changes in the intensity of policing on crime with a distinct focus on the crime-reducing effect of various "best practices." In particular, declines in crime that are not attributable to spatial displacement have been linked to the adoption of "hot spots" policing (Sherman and Rogan 1995, Sherman and Weisburd 1995, Braga 2001, Braga 2005, Weisburd 2005, Braga and Bond 2008, and Berk and MacDonald 2010), "problem-oriented" policing (Braga et al. 1999; Braga et al. 2001; and Weisburd et al. 2010), and a variety of other proactive approaches. Similarly, a large research literature that has examined the local impact of police crackdowns has consistently found large and immediate (but typically not lasting) reductions in crime in the aftermath of hyper-intensive policing (Sherman 1990). Such findings are further supported by evidence from several informative natural experiments that have identified plausibly exogenous variation in the intensity of policing. Three prominent examples are Klick and Tabarrok (2005), who study the effect of police redeployments in Washington, DC, that result from shifts in terror alert levels:

Di Tella and Schargrodsky (2004), who study the effect of a shift in the intensity of policing in certain areas of Buenos Aires after a 1994 synagogue bombing; and Draca, Machin, and Witt (2011), who study police redeployments in the aftermath of the 2005 London tube bombings.

The literature on police deployments and tactics has focused predominantly on three types of interventions. The first is an innovation commonly referred to as "hot-spots" policing. As the moniker suggests, hotspots policing describes a strategy in which police are disproportionately deployed to areas in a city that appear to attract disproportionate levels of crime.²⁰ The second type of intervention is often referred to as 'problem-oriented" policing. This term is used broadly and refers to a collection of focused deterrence strategies that are designed to change the behavior of specific types of offenders or to be successful in specific jurisdictions. A final intervention that has received attention in the literature is that of "proactive" policing. Proactive policing refers to strategies that are deigned to make policing more intensive, holding resources fixed. The idea can be traced back to the concept of "broken windows," or disorder policing introduced by Wilson and Kelling (1982) and refers to the notion that, just like fixing a broken window sends a message to would-be vandals that the community cares about maintaining social order, arresting individuals for relatively minor infractions sends a message to potential offenders that the police are watchful.

²⁰The idea that crime hot spots might exist is immediately obvious to many and can be found in the academic literature at least as far back as Shaw and McKay (1942). Modern research has linked criminal activity to specific types of places such as bars (Roman and Reid 2012) and apartment buildings, as well as to places that lack formal or informal guardians (Eck and Weisburd 1995).

3.2.1 Hot-Spots Policing

We begin with a discussion of hot-spots policing, which we distinguish from aggregate police manpower research for several reasons. First, as the manpower literature largely uses city-level variation, it identifies the effect of adding police at the expense of some other type of public input. In contrast, hot-spots policing involves a reallocation of existing resources. Such a strategy is advantageous, as it does not require a change in current outlays. However, it also leaves open the possibility that moving police around merely shifts, rather than reduces, crime.

In order for hot-spots policing to be a viable crime reduction strategy, two conditions must be met. First, given resource constraints, the feasibility of such a deployment strategy relies on crime being sufficiently concentrated in a relatively small number of hot spots. Second, hot spots must be sufficiently stable such that the spatial distribution of crime in the absence of a change in police deployment can be predicted with a reasonable degree of accuracy. Hence, the adoption of hot-spots policing must begin with an accounting of the geographic concentration of crime, as well as an assessment of the extent to which hot spots are permanent as opposed to transitory. Sherman (1995) captures both of these ideas, characterizing crime hot spots as "small places in which the occurrence of crime is so frequent that it is highly predictable, at least over a one-year period."

A seminal paper by Sherman, Gartin, and Buerger (1989) is the first to provide descriptive data on the degree to which crime is spatially concentrated. Using data from Minneapolis, Sherman and coauthors found that just 3 percent of addresses and intersections in Minneapolis produced 50 percent of all calls for service to the police. This finding is echoed by Weisburd, Maher, and Sherman (1992) and in more recent papers by Weisburd et al. (2004) and Weisburd, Morris, and Groff (2009), which report that a very small percentage of street segments in Seattle accounted for 50 percent of crime incidents for each year over a fourteen-year period.²¹ With respect to predictability, Weisburd et al. (2004), using the same data from Seattle, used trajectory analysis to establish that hot spots tended to be highly persistent, often persisting for many years.²²

Naturally, the observation that crime is so highly concentrated in a very small number of places has led to efforts to intensify the focus of police resources on these places. These interventions have, in turn, led to a corresponding experimental and quasi-experimental research literature that seeks to evaluate the efficacy of such strategies. The first-order question that the hotspots policing literature seeks to address involves the degree to which highly localized crime is responsive to a change in the intensity of policing. By responsive, criminologists generally refer to the idea that crime declines in local areas that have been exposed to more intensive patrol without merely inducing equivalent spillovers to untreated adjacent areas. However, we note that while spillovers undermine the viability of hot-spots policing as a crime-reduction strategy, they nevertheless constitute evidence of responsiveness and, as such, are useful in identifying deterrence. Moreover, a particular feature of this research makes it especially salient for the study of deterrence (Nagin 2013). Notably, while the literature tends to find that intensive policing reduces crime, elements of intensive policing such as rapid response times do not appear to increase the likelihood of an arrest (Spelman

²¹An excellent review of this literature may be found in Weisburd, Bruinsma, and Bernasco (2009).

²² Hot spots can, of course, also be temporary. An excellent accounting of efforts to predict temporary hot spots in Pittsburgh can be found in Gorr and Lee (2015).

and Brown 1981). Such a pattern in the data tends to be consistent with deterrence but not with incapacitation.

The first test of policing crime hot spots may be found in a 1995 randomized experiment conducted by Sherman and Weisburd in Minneapolis. The experiment tested whether doubling the intensity of police patrols in crime hot spots resulted in a decrease in crime and found that crime declined by approximately 10 percent in experimental places relative to control places. No evidence of crime displacement—that is, spillovers-was found. Findings in Sherman and Weisburd (1995) have, to a large extent, been replicated in other places and contexts including the presence of open-air drug markets and "crack houses" (Hope 1994, Weisburd and Green 1995, and Sherman et al. 1995), violent crime hot spots (Sherman and Rogan 1995, Braga et al. 1999, and Caeti 1999), and places associated with substantial social disorder (Braga and Bond 2008 and Berk and MacDonald 2010). Indeed, a review of the literature by Braga (2001) identified nine experiments or quasi-experiments involving hot-spots policing and noted that seven of the nine studies, including a majority of the randomized experiments, found evidence of significant and large crime reductions. Notably, a majority of the literature finds no evidence of displacement of crime to adjacent neighborhoods (Weisburd et al. 2006), while a number of studies have found that the opposite is true-that there tends to be a diffusion of benefits to nontreated adjacent places (Sherman and Rogan 1995, Braga et al. 1999, and Caeti 1999).²³ Both of these findings are perfectly consistent with our conceptualization of deterrence.

3.2.2 Problem-Oriented Policing

Intensive policing of hot spots is one way that police potentially deter crime. Another broad deterrence-based strategy is that of problem-oriented policing. Broadly speaking, this strategy entails engaging with community residents to identify the most salient local crime problems and designing strategies to deter unwanted behavior. The specifics are highly variable by design and are intended to leverage local resources to address highly local concerns. What these strategies have in common and why they are frequently referred to as "focused deterrence" strategies is that each of them attempts to generate deterrence through advertising (Zimring and Hawkins 1973). The idea is to create deterrence by making potential offenders explicitly aware of the risks of serious criminal involvement.

Undoubtedly the most well-known evaluation of a problem-oriented policing approach is that of Boston's Operation Ceasefire by Kennedy et al. (2001). The stated purpose of Ceasefire was to reduce youth gun violence in Boston. The intervention involved a multifaceted approach and included efforts to disrupt the supply of illegal weapons to Massachusetts. It also included messages communicated by police directly to gang members that authorities would use every available "lever" to punish gangs collectively for violent acts committed by individual gang members. In particular, police indicated that the stringency of drug enforcement would hinge on the degree to which gangs used violence to settle business disputes. The result of the intervention was that youth violence fell considerably in Boston relative to other US cities included in the study.

Indeed, the perception of Ceasefire has been overwhelmingly positive and accordingly it has given rise to a number of similarly motivated strategies that are collectively referred to as "pulling levers."

 $^{^{23}}$ An excellent review of the theory and empirical findings regarding displacement in this literature can be found in Weisburd et al. (2006).

Prominent evaluations of pulling-levers interventions include research carried out in Richmond, VA (Raphael and Ludwig 2003), Indianapolis (McGarrell et al. 2006), Chicago (Papachristos, Meares, and Fagan 2007), Stockton, CA (Braga 2008b), Lowell, MA (Braga et al. 2008), High Point, NC (Corsaro et al. 2012), Nashville (Corsaro and McGarrell 2010), Cincinnati (Engel, Corsaro, and Tillyer 2010), and Rockford, IL (Corsaro, Brunson, and McGarrell 2010). Researchers have also evaluated a multicity pulling-levers strategy known as Project Safe Neighborhoods (PSN), which enlisted the cooperation of federal prosecutors to crack down on gun violence. A 2012 review of the literature by Braga and Weisburd suggests that pulling-levers strategies have been effective in reducing serious violent crime, with all reviewed studies finding negative point estimates, the majority of which were significant. With respect to individual evaluations, reductions in crime have been found in High Point, Chicago, Indianapolis, Stockton, Lowell, Nashville, and Rockford and null findings have been found in Richmond and Cincinnati.²⁴ With respect to Project Safe Neighborhoods, the research is promising but not definitive. McGarrell et al. (2010) report that declines in crime were greater in PSN cities than in non-PSN cities. However, there is a great deal of heterogeneity among cities, making it difficult to draw clear inferences.

On the whole, evaluations of pullinglevers strategies produce promising results, though inference is invariably complicated by a lack of randomized experiments and the inherent difficulty in identifying appropriate comparison cities. Identification problems are additionally compounded by the difficulty in identifying mechanisms, as each pulling-levers strategy is complex, multifaceted, and situation dependent, often involving changes in both the intensity of law enforcement as well as sentencing (e.g., Project Exile in Richmond, VA, as well as Project Safe Neighborhoods). Accordingly, it is easy to imagine that as additional resources are brought to bear, some of the effects of pulling-levers strategies might accrue via incapacitation effects. Concerns regarding identification have led Skogan and Frydl (2004) to conclude that such research is "descriptive rather than evaluative." Given the relatively large effect sizes reported in the literature, our reading of these papers is more optimistic than that of Skogan and Frydl. However, caution is warranted in characterizing this literature as having detected unassailable evidence of deterrence.

3.2.3 Proactive and Disorder Policing

A final strand of the police tactics literature in criminology investigates the responsiveness of crime to the intensity of policing, holding resources constant, an idea that is generally referred to as "proactive" policing. As there is no standardized way to assess the extent to which individual police departments engage in police work that is proactive, in practice, this literature seeks to understand if the intensity of arrests for minor infractions has an effect on the incidence of more serious crimes. Building on a proposition in Wilson and Kelling (1982), such an empirical operationalization was first proposed by Sampson and Cohen (1988) and has been replicated to various degrees by MacDonald (2002) and Kubrin et al. (2010). The general strategy is to regress crime rates on a measure of policing intensity. In practice, policing intensity has been operationalized using the number of driving under the influence (DUI) and disorderly conduct arrests made per police officer. Using this approach has, in some cases, led to findings that are consistent with a deterrence effect of proactive

 $^{^{24}}$ Braga and Weisburd (2012) provided an excellent review of the literature including a comprehensive meta-analysis of the research findings.

policing. However, in the best controlled models, coefficients on the proactive policing proxy become small and insignificant. More importantly, these models are plagued by problems of simultaneity bias, omitted variables, and the inevitable difficulty involved in finding a credible proxy for the concept of proactive policing, as opposed to simply an environment that is rich in opportunities for police officers to make arrests.

A second focus of the literature has been on the advent of broken-windows policing (also known as "order maintenance" or "disorder" policing, a policy innovation proposed by Wilson and Kelling 1982). The idea behind broken-windows policing is that police can affect crime through tough enforcement of laws governing relatively minor infractions such as vandalism and turnstile jumping. Broken-windows policing, in theory, operates primarily through perceptual deterrence-if offenders observe that police are especially watchful, they may update their perceived probability of apprehension for a more serious crime and accordingly will decrease their participation in crime. In the popular media, broken-windows policing is an idea that is heavily associated with Mayor Rudolph Giuliani and New York Police Commissioner William J. Bratton, who has attributed the dramatic decline in crime in New York City after 1990 to its rollout (Kelling and Bratton 2015).

A corresponding research literature has arisen to evaluate the effectiveness of broken-windows policing—in practice, this literature has focused disproportionately on the experience of New York City, which experienced the largest decline in crime among major US cities. This literature produces mixed findings. On the one hand, time series analyses by Kelling and Sousa (2001) and Corman and Mocan (2005) find that misdemeanor arrests are negatively associated with future arrests for more serious crimes such as robbery and motor vehicle theft. On the other hand, later research has pointed out that these studies omit a control group and has tended to focus on the fact that New York's aggregate crime trends, while steeper, are broadly similar to those of other cities that did not institute a policy of broken-windows policing (Eck and Maguire 2000; Rosenfeld, Fornango, and Baumer 2005; and Harcourt and Ludwig 2006). The two most credible analyses, those of Harcourt and Ludwig (2006) and Rosenfeld, and Rengifo (2007), Fornango, use precinct-level data on misdemeanor arrests and violations and find either no effect or very small effects. More fundamentally, there are a number of alternative explanations for New York's dramatic reduction in crime including the receding of the crack epidemic (Blumstein 1995), changes in demographics that are poorly measured at lower levels of geographic granularity, general strategies to address disorder such as boarding abandoned buildings, and the implementation of the data-driven Compstat system (Weisburd et al. 2003). Accordingly, even if identification problems can be set aside, it is unclear that this literature can isolate the impact of disorder policing from other changes that drove crime down in New York City. Given the dramatic rollback of the New York City Police Department's "stop-and-frisk" policy in 2014 and the continued decline in serious crime, as well as the failure of the most careful studies to find evidence of large effects, we are skeptical that disorder policing has played a large role in the decline in crime in New York City.²⁵ Overall, our reading of this literature is that the evidence in favor of an important effect of proactive policing on crime is weak.

²⁵Broken-windows policing and the associated stopand-frisk policy implemented by the New York City police department has generated substantial public controversy. A 2009 paper by Fagan et al. (2009) provides evidence of the demographic burden of such policies that is disproportionately borne by African Americans.

Of course, New York City is not the only municipality to experiment with disorder policing and, in our view, some of the strongest evidence can be found in research from other cities. Three papers that employ especially strong research designs are worth mentioning. Braga et al. (1999) provides the first experimental evaluation of a strategy designed explicitly to address disorder. In Jersey City, NJ, twelve of twenty-four crime hot spots were randomly assigned to receive an intervention that involved disorder policing as well as other place-specific treatments that were intended to reduce crime. Such treatments include clearing vacant lots, requiring store owners to clean store fronts, and facilitating more frequent trash removal. Treated places experienced large declines in both crime and calls for service. In a follow-up study in Lowell, MA, Braga and Bond (2008) attempted to further isolate disorder policing from other types of disorder reduction, randomly assigning seventeen Lowell hot spots to receive a general disorder policing strategy. This study also showed strong reductions in crime in treated areas. However, the greatest gains were found in areas with an especially heavy focus on situational crime prevention, as opposed to arresting larger numbers of low-level offenders. Evidence in favor of an effect of misdemeanor arrests is far more limited. Finally, a particularly careful paper by Caetano and Maheshri (2014) finds no evidence of an effect of "zero tolerance" law enforcement policies on crime using microdata from police precincts in Dallas. Taken as a whole, the evidence suggests that reducing disorder is a promising strategy for controlling crime. However, it is difficult to characterize these reductions as deterrence. In particular, disorder reduction may simply help people to feel better about their neighborhoods, thus representing a shift in preferences, rather than movement along the curve that is induced by an increase in the perceived probability

of capture by police. We remain skeptical that disorder *policing* provides evidence of deterrence.

3.2.4 Changes in City-Wide Police Deployments

The ubiquity of the hot spots, problemoriented and proactive policing literatures in criminology has spawned a parallel literature in economics that seeks to learn from natural experiments in police deployments. This literature is conceptually similar to the hot-spots literature with two exceptions. First, the identifying variation is naturally occurring in contrast to experimental manipulation, which may be excessively contrived. Second, several of the natural experiments identify the impact of a diffuse reduction in resources, rather than a concentration of resources at particular hot spots.

Three prominent studies are those of Di Tella and Schargrodsky (2004); Klick and Tabarrok (2005); and Draca, Machin, and Witt (2011). Each of these studies leverages a redeployment of police in response to a perceived terrorist threat. The appeal of these studies is that terrorist threats are plausibly exogenous with respect to trends in city-level crime and therefore represent a unique opportunity to learn about the response of crime to changes in normal routines of policing. Di Tella and Schargrodsky study the response of police in Buenos Aires to the 1994 bombing of a Jewish community center. In the aftermath of the bombing, Argentine police engaged in a strategy of "target hardening" synagogues by deploying disproportionate numbers of officers to blocks with synagogues or other buildings housing Jewish organizations. Di Tella and Schargordsky report that the intervention led to a large decline in motor vehicle thefts on the blocks that received additional police patrols though the effects. Notably, this result has been called into question by Donohue, Ho, and Leahy (2013), who reanalyzed

the original data and report evidence that is more consistent with spatial displacement of crime rather than crime reduction. However, with respect to identifying behavioral changes among offenders, both stories are equally consistent with deterrence. In a similar study, Klick and Tabarrok (2005) utilize the fact that when terror alert levels set by the US Department of Homeland Security rise, property crime (but not violent crime) tends to fall in Washington DC, with especially large declines in areas that receive the largest redeployments of police protection. With respect to the United Kingdom, Draca, Machin, and Witt (2011) study the 2005 London tube bombings which resulted in sizable shifts in the deployments of police from the suburbs to central London and find that "street crimes" such as robbery and theft are reduced considerably in areas that received additional officers.

With respect to studying variation in the spatial concentration of police, two additional papers are worth noting. Cohen and Ludwig (2003) exploit short-term variation in the intensity of police patrols by day of the week in several different Pittsburgh patrol areas. They found that shootings were considerably lower in areas and on days that received more intensive police patrols. With respect to the long-term consequences of patterns of police deployments, MacDonald, Klick, and Grunwald (2016) use a spatial regression discontinuity (RD) design to study the impact of especially intensive policing around the University of Pennsylvania, a large urban university campus. In particular, areas directly adjacent to the university received police patrols from both the university and municipal police. Areas slightly further away from the campus received only municipal police patrols. The finding is that street crimes are substantially higher in the blocks just outside the area patrolled by the university police relative to the blocks just inside the university patrol area.

3.3 Deterrence versus Incapacitation

The literature has reached a consensus that increases in police manpower reduce crime, at least for a population-weighted average of US cities. With respect to police deployments and tactics, the literature supports the idea that crime is responsive to a visible police presence in hot spots and pulling-levers strategies that advertise deterrence, while evidence in favor of an effect of proactive policing strategies such as broken windows and disorder policing is more suspect. A remaining issue is to address the degree to which each of these literatures is informative with respect to disentangling deterrence from incapacitation.

With respect to the aggregate manpower literature, Levitt (1998) provides the first attempt to systematically unpack the relationship between deterrence and incapacitation by empirically examining the link between arrest rates and crime, a relationship that is negative. Levitt posits that this negative relationship can be explained either by deterrence, incapacitation, or measurement errors in crime. Ruling out measurement errors as a likely culprit, he differentiates between deterrence and incapacitation using the effect of changes in the arrest rate for one crime on the rate of other crimes.²⁶ As Levitt notes, "in contrast to the effect of increased arrests for one crime on the commission of that crime, where deterrence and incapacitation are indistinguishable, it is demonstrated that these two forces act in opposite directions when looking across crimes. Incapacitation suggests that an increase in the arrest rate for one crime will reduce all crime rates;

²⁶Utilizing an insight from Grilliches and Hausman (1986)—that measurement errors should yield the greatest bias in short-differenced regressions—Levitt (1998) compares regression estimates of the relationship between crime and arrest rates using short- and long-differences, finding similar effects.

deterrence predicts that an increase in the arrest rate for one crime will lead to a rise in other crimes as criminals substitute away from the first crime." Levitt concludes that deterrence appears to be the more important factor, particularly for property crimes. Owens (2013) reports a similar finding, examining whether variation in police staffing resulting from the COPS hiring program led to increased arrests. Despite the fact that the program, which provided funding to increase the number of patrol officers in US cities, appears to have led to a decline in crime, no significant effect is found on arrests. As a result, Owens concludes that there is little evidence in favor of incapacitation, which necessarily must operate through arrests, thus implying a large role for deterrence.

While analyses by Levitt (1998) and Owens (2013) are suggestive of a meaningful role for manpower-induced deterrence, it is nevertheless difficult to disentangle deterrence from incapacitation in this way. In particular, a null relationship between police and arrests is also consistent with the idea that police productivity decreases when there are fewer crimes to investigate. Moreover, the imprecise parameter estimates on arrest along with standard errors that are not trivial in size in Owens (2013) render it difficult to make strong claims regarding the null effect of police on arrests. For this reason, while the aggregate-data literature is ideal for understanding the overall relationship between police and crime, it is only somewhat informative with respect to the magnitude of deterrence. This point is further compounded by the observation that research has yet to document the degree to which offenders perceive or are aware of increases in police manpower (Nagin 1998).

We suspect that the literature on police tactics is considerably more informative with respect to identifying deterrence. In particular, offenders are more likely to be aware of an enhanced police presence in small, local areas than relatively small changes in the number of police in a city spread out over a large geographic area. Likewise, while offenders tend to commit crimes locally, in order for incapacitation to explain the large declines in crime that occur in hot spots, it would have to be the case that offending is so local so as to be specific to a group of one or two blocks. The large drops in crime that occur in crime hot spots after they are more aggressively policed is more consistent with deterrence than with incapacitation. Focused deterrence strategies are also particularly informative in that declines in crime have been shown to be specific to the focus of the intervention. To the extent that at least some offenders are generalists, rather than specialists who commit only a certain type of offense, such a pattern is more consistent with deterrence than with incapacitation.

In sum, while it remains possible that an increased police presence lowers crime by situating police officers in locations where they are more likely to arrest and incapacitate potential offenders, on the whole, the high degree of visibility around police crack-downs or hot spots policing suggests a potentially greater role for deterrence.²⁷

4. Sanctions and Crime

A second idea in Becker's neoclassical model of offending is that crime will be responsive to the certainty and severity

²⁷An important exception to this intuition, however, can be found in Mastrobuoni (2013), who studies the responsiveness of crime to regular shift changes among the various police forces in Milan. Matsrobuoni finds that despite large temporal discontinuities in clearance rates during shift changes, robbers do not appear to exploit these opportunities and concludes that there is only limited evidence of deterrence. A remaining question is the extent to which the result depends on the ability of potential offenders to accurately perceive these discontinuities.

of punishment.²⁸ Accordingly, a parallel literature considers the responsiveness of crime to the harshness of criminal sanctions, along both the intensive and extensive margin. Three literatures, in particular, are worth mentioning. First, a series of papers considers the effect of sentencing policy generally or, alternatively, sentence enhancements on crime, to test the prediction that crime will decrease in response to a sanction regime that either raises the probability of a prison sentence or raises the length of a prison sentence, if given. In practice, this literature focuses primarily on the intensive margin, that is, the severity of punishment rather than the probability that a custodial punishment is given conditional upon being arrested. A corresponding literature considers the effect of laws that govern the age of criminal majority and, as such, generate large and pervasive discontinuities in the sanctions that individual offenders face. Since adult sanctions are more intensive along both the intensive and extensive margins, such studies identify a reduced-form deterrence effect that does not explicitly differentiate between the certainty and the severity of punishment. Finally, a particularly prominent literature considers the effect of a capital-punishment regime or the incidence of executions on murder. Since executions enhance the expected severity of the sanction without directly affecting an offender's probability of capture, this literature is potentially compelling with respect to understanding deterrence as, subject to satisfying the standards of econometric identification, it allows for the isolation of a pure deterrence effect operationalized along the intensive margin.

4.1 Sentencing

One of the most basic tests of the Becker model of crime concerns the responsiveness of crime to the harshness of criminal sanctions. Over the past few decades, a literature has arisen to document the sensitivity of crime to various sentencing schemes, sentence enhancements, clemency policies, "three strikes" laws, and other legislative actions that change the expected cost of a criminal sanction. A corresponding literature measures the responsiveness of crime to the size of the prison population. With respect to identification, two challenges are particularly pressing. First, it is difficult to discern the effect of sentencing policies (which, in the United States, are generally enacted at the state level) from other crime reduction interventions, as well as time-varying factors that inform the supply of crime more generally. Attempts to isolate the causal effects of a change in state-level sentencing policy invariably encounter the inevitably difficult issues of choosing an appropriate comparison group and selecting from among many competing and equally plausible models of aggregate offending. Durlauf and Nagin (2011) refer to the latter of these issues as the problem of ad hoc model specification, referring specifically to the under-theorized manner in which individual-level mental processes are modeled and the arbitrary choice of control variables in regressions.

Second, just as prison populations may affect crime, crime may have a reciprocal effect on prison populations, creating the potential for simultaneity bias.²⁹ With respect to identifying deterrence, the chief difficulty is that harsher sanctions may lead to deterrence, but typically also to incapacitation.

²⁸The term "certainty of punishment" is often used in the literature to refer either to the probability that an individual is apprehended or to the overall probability that an individual is punished conditional upon offending. In this section, in referring to the certainty of punishment, we are focusing more specifically on the probability that a punishment is handed out conditional upon arrest. This refers to the severity of punishment along the extensive margin.

²⁹For a comprehensive review of identification issues in this literature, see Durlauf and Nagin (2011).

This section reviews the literature that seeks to understand the relationship between sanctions and offending with a particular interest in discerning the effect that sanctions have on deterrence.

4.1.1 Prison Populations and Crime

While identifying the elasticity of crime with respect to a sanction, in principle, requires an exogenous shock to the sanctions regime, a natural starting point in unraveling the crime-sanctions relationship is to consider the elasticity of crime with respect to the size of the prison population. Studies of the crime-prison population elasticity generally utilize state-level panel data and regress the growth rate in crime on the first lag of the growth rate in a state's share of prisoners. Marvell and Moody (1994) provide the first credible empirical investigation of the elasticity of crime with respect to prison populations, estimating an elasticity of -0.16. As in their police paper, they use the concept of Granger causality in an attempt to rule out a causal relationship that runs from crime to prison populations. As discussed in section 2, the approach, while useful, does not offer compelling reasons to believe in the ignorability of selection bias.

A genuine breakthrough in this literature is found in Levitt (1996) who, using similar data, exploits exogenous variation in state incarceration rates induced by court orders to reduce prison populations. The intuition behind the approach is that the timing of discrete reductions in a state's prison population owing to a court order should be as good as random. This may not be strictly true, as the necessity of court orders to reduce overcrowding may itself be a function of rising crime rates. However, the strategy relies more specifically on the randomness of the precise timing of the order and, in our judgment, represents a plausible strategy for identifying a causal estimate of the effect of prison populations on crime. Levitt's estimated elasticities are considerably larger than those in Marvell and Moody: -0.4 for violent crimes and -0.3 for property crimes, while the largest elasticity reported is for robbery (-0.7).³⁰ An alternative identification strategy can be found in Johnson and Raphael (2012), who develop an instrumental variable to predict future changes in incarceration rates. The instrument is constructed by computing a theoretically predicted dynamic adjustment path of the aggregate incarceration rate in response to a given shock to prison entrance and exit transition probabilities. Given that incarceration rates adjust to permanent changes in behavior with a dynamic lag, the authors identify variation in incarceration that is not due to contemporaneous criminal offending. Using state-level panel data covering 1978–2004, Johnson and Raphael (2012) estimate the elasticity of crime with respect to prison populations of approximately -0.1 for violent crimes and -0.2 for property crimes. Notably, the estimated elasticities in Johnson and Raphael for earlier time periods were considerably larger and closer in magnitude to those estimated by Levitt (1996). Johnson and Raphael conclude that the criminal productivity of the marginal offender has changed considerably over time as incarceration rates have risen, a conclusion that is echoed by Liedka, Piehl, and Useem (2006). With respect to juveniles, Levitt (1998) studies the response of juvenile crime to the punitiveness of state-level juvenile sentencing along the extensive margin (the number of juveniles in custody per capita), concluding that changes in juvenile sentencing explain approximately 60 percent of the growth in juvenile crime during the 1970s and 1980s. Using Levitt's results, Lee and McCrary (forthcoming) compute an implied elasticity for violent crimes of -0.4.

³⁰Levitt's analysis is replicated by Spelman (2000) who reports qualitatively similar findings.

In sum, estimates of the elasticity of crime with respect to prison are generally modest in comparison to police elasticities and fall between -0.1 and -0.7. The most recent estimates fall in the low end of that range. Estimates for violent and property crimes are of approximately equal magnitude and there is evidence that the elasticity has diminished considerably over time as prison populations have grown. Our best guess is that the current elasticity of crime with respect to prison populations is approximately -0.2, as reported by Johnson and Raphael (2012).31 This finding is further bolstered by a recent evaluation of "realignment," a policy implemented in California to reduce prison overcrowding by sending additional inmates to county jails, where they tend to serve shorter sentences. Lofstrom and Raphael (2013) report that, with the exception of motor vehicle theft, there is no evidence of an increase in crime despite the fact that 18,000 offenders who would have been incarcerated are on the street due to the realignment policy. The magnitude of this elasticity leaves open the possibility for nontrivial deterrence effects of prison but, given that prison generates sizable incapacitation effects, the magnitude of deterrence is likely small.

4.1.2 Shocks to the Sanctions Regime

A related literature considers the effect of a discrete change in a jurisdiction's sanctions regime that is plausibly not a function of crime trends more generally. The general approach is to utilize a differences-in-differences design to compare the time-path of crimes covered by a sentence enhancement to that of uncovered crimes. The earliest literature (Loftin and McDowall 1981; Loftin, Heumann, and McDowall 1983; Loftin and McDowall 1984; and McDowall, Loftin, and Wiersma 1992) considered the effects of sentence enhancements for specific crimes—particularly gun crimes—generally finding little evidence in favor of deterrence. A more recent paper studies the impact of changes in sentencing in the aftermath of London's 2011 riots. Leveraging the fact that judges in the United Kingdom handed down harsher sentences for "riot offenses" in the six months following the riots, Bell, Jaitman, and Machin (2014) find evidence of sizable declines in riot offenses relative to nonriot offenses which, in the absence of identifiable changes in policing, they attribute to the advent of a harsher sanctions regime. This claim is bolstered by the fact that there was a relative decline in riot offenses in sectors that experienced the brunt of the 2011 riots, as well as sectors that saw no riot activity.³²

A second class of studies has examined the impact of changes in the sanctions regime that have heterogeneous impacts on different groups of offenders. For example, Drago, Galbiati, and Vertova (2009) study the effect of a 2006 collective clemency of incarcerated prisoners in Italy. Prisoners incarcerated prior to May 2006 were released from prison with the remainder of their sentences suspended, while prisoners incarcerated after May 2006 were ineligible for the clemency. Released prisoners, however, were subject to a sentence enhancement for any future crimes committed that were serious enough to merit a sentence of at least two vears. For such crimes, the sentence would be augmented by adding the amount of time the prisoner was sentenced to serve prior to his pardon to his new sentence. Thus, the intervention created a situation in which otherwise similar individuals convicted of the same crime faced dramatically different sanctions regimes. The results of this natural

 $^{^{31}\}mathrm{A}$ 2009 review of the literature by Donohue reaches a similar conclusion.

³²Sentencing did change along both the intensive and extensive margins, indicating the incapacitation cannot be ruled out.

experiment suggest an elasticity of crime with respect to sentence length of approximately -0.5 at one year follow-up. Utilizing the same natural experiment, Buonanno and Raphael (2013) report evidence that incapacitation effects forgone as a result of the collective clemency were large, thus constraining the magnitude of the deterrence effect.

Similar findings are reported for the United States by Helland and Tabarrok (2007). Using data from California's threestrikes regime, Helland and Tabarrok (2007) compare the criminal behavior of individuals convicted of a second "strikeable" offense to those tried for a second strikeable offense but who were ultimately convicted of a lesser offense. As Durlauf and Nagin (2011) note, individuals with one strike may not be an ideal comparison group for a variety of reasons—in particular, it may be the case that the individuals with two strikes had poorer legal representation or that the precise nature of their potential second-strike offense was qualitatively less serious. Nevertheless, the authors demonstrate that comparing twostrike to one-strike individuals is sufficient to remove a great deal of the selection bias that exists in comparing individuals with two strikes to the remainder of the charged population. The authors find evidence of an appreciable deterrent effect, calculating that California's three-strikes legislation reduced felony arrest rates by approximately 20 percent among criminals with two strikeable offenses against them. Similarly, while Zimring, Hawkins, and Kamin (2001) find little evidence of an overall effect of threestrikes legislation, they do find evidence that individual offenders with two strikes are less likely to be arrested. Given that the deterrence margin is most salient at two strikes, these studies stand out as especially important with respect to identifying a meaningful deterrence effect of sentencing. On the other hand, the magnitude of the response

is actually quite small once one considers the increase in sentence lengths associated with three strikes. Helland and Tabarrok's estimates suggest an elasticity of crime with respect to sentence length of -0.06.

Last but not least, we survey a completely different idea with respect to changing the sanctions regime. While sentence enhancements and three-strikes laws are designed specifically to increase sanction severity across either the intensive or the extensive margin or both, it is possible to imagine simultaneously making one margin harsher and the other one less harsh. This is the premise underlying swift-and-certain sanctions regimes (Hawken and Kleiman 2009 and Kleiman 2009). The idea of swift-andcertain sanctions arises from the notion that myopic individuals are unlikely to be responsive to long sentences but may be highly responsive to short sentences if they are issued with near certainty. In recent practice, Hawaii's Opportunity Probation with Enforcement (HOPE) program is the canonical example of swift-and-certain sanctioning in action. In an effort to address chronic recidivism among probationers, Hawaii First Circuit Court Judge Steven Alm recognized that punishments for violating the terms of probation were fairly unlikely and, if meted out, tended to occur in the distant future. Moreover, the sanctions were typically harsh and, as such, costly. Judge Alm and his collaborators put into practice a program that addressed probation violations with immediate but light sanctions-typically ranging from warnings to spending up to a week in jail. Probationers were intensively monitored, with any violations resulting in a sanction. In a banner finding, Hawken and Kleiman (2009) find that individuals assigned at random to HOPE, as opposed to business as usual, were 55 percent less likely to be arrested for a new crime, 72 percent less likely to use drugs, and 53 percent less likely to have their probation revoked than

those on regular probation. In a similarly promising related study, Kilmer et al. (2013) found that a swift-and-certain program in South Dakota targeted towards persistent alcohol-involved offenders appears to have had extraordinarily large effects in counties that received the program.

4.2 Capital Punishment Regimes

Variation in the presence or intensity of capital punishment generates a potentially excellent source of variation with which to test for the magnitude of general deterrence. In particular, to the extent that variation in a state's capital-punishment regime is unrelated to changes in the intensity of policing, the effect of capital punishment represents a pure measure of deterrence with any response of murder to the presence or intensity of capital punishment not plausibly attributable to incapacitation.³³

There have been two primary approaches to identifying deterrence effects of capital punishment. One approach considers the use of granular time-series data or event studies to identify the effect of the timing of executions on murder. Time-series studies typically use vector autoregression to assess whether murder rates appear to decline in the immediate aftermath of an execution. Prominent examples include Stolzenberg and D'Alessio (2004), which finds no evidence of deterrence, and Land, Teske, and Zheng (2009), which finds evidence of short-run deterrence. Event studies such as those of Grogger (1991) and Hjalmarsson (2009a) examine the daily incidence of homicides before and after executions. Both Grogger (1991) and Hjalmarsson (2009a) find little evidence of deterrence effects though, as Charles and Durlauf (2013) and Hjalmarsson (2012) note, with a limited time horizon, it is not possible to distinguish between what we typically think of as deterrence and temporal displacement. A related study, Cochran, Chamlin, and Seth (1994), considers the effect of Oklahoma's first execution in more than twenty years and finds evidence that the execution appears to have increased murder among strangers, an effect they attribute to a "brutalization" hypothesis, though it is attributed with equal ease to statistical noise. A final study worth noting is that of Zimring, Fagan, and Johnson (2010), who compare homicide rates between Singapore, which uses the death penalty with variable intensity, and Hong Kong, which does not use the death penalty. The paper finds no evidence in favor of deterrence, as both countries experience similar homicide trends over the thirty-five-year time period studied.

Broadly speaking, the time-series and event-studies literatures offer little support in favor of deterrence though, as noted by Charles and Durlauf (2013), the literature is plagued by several conceptual problems that compromise the interpretability of estimated treatment effects. In particular, the focus of the time-series literature on executions, as opposed to the sanctions regime more generally, marks a divergence from the neoclassical model of crime insofar as the occurrence of an execution does not per se change the expected severity of a criminal sanction for murder.³⁴ Indeed the research design is often motivated by the assumption that an execution affects an offender's perceived

³³The argument is that in the absence of a capital punishment regime or a death sentence, a convicted offender would nevertheless be sentenced to a lengthy prison sentence (such as a life sentence) without the possibility of parole.

³⁴An important exception to this general point can be found in Chen (2013), which studies the effect of executions for desertion among British soldiers during World War I and finds evidence that executions deter desertion, but actually encourage desertion when the execution was for an offense other than desertion or if the executed soldier was Irish. The reason this study stands as an exception to the rule proposed by Charles and Durlauf is that during a time of war, the sanction regime is likely to be in constant flux.

sanction. However, there is little evidence, empirical or otherwise, to support this assumption. Second, Charles and Durlauf note that the underlying logic of time-series analyses of executions and murder operationalize as deterrence the dynamic correlations between a shock to one time series and the levels of another. As the authors note, this is an arbitrary conceptualization of what is meant by deterrence.

A second literature studies the deterrent effect of capital punishment utilizing panel data on US states to identify the effect of a capital-punishment statute or the frequency of executions on murder among the public at large. In particular, these studies have exploited the fact that in addition to cross-state differences in sentencing policy, there is also variation over time for individual states in the official sentencing regime, the propensity to seek the death penalty in practice, and the application of the ultimate punishment (Chalfin, Haviland, and Raphael 2013). This literature has generated mixed findings with several prominent papers (e.g., Dezhbakhsh, Rubin, and Shepherd 2003; Mocan and Gittings 2003; Zimmerman 2004, 2006; and Dezhbakhsh and Shepherd 2006) finding large and significant deterrence effects, and several equally prominent papers (Katz, Levitt, and Shustorovich 2003; Berk 2005; Donohue and Wolfers 2005, 2009; and Kovandzic, Vieraitis, and Paquette-Boots 2009) finding little evidence in favor of deterrence.35

While evidence in favor of deterrence is mixed, recent reviews by Donohue and Wolfers (2005, 2009) and Chalfin, Haviland, and Raphael (2013), as well as a 2011 report commissioned by the National Academy of Sciences, point to substantial problems in a

number of papers that purport to find deterrence effects of capital punishment. These problems include the use of weak and/or inappropriate instruments (Dezhbakhsh, Rubin, and Shepherd 2003; and Zimmerman 2004), failure to report standard errors that are robust to within-state dependence (Dezhbakhsh and Shepherd 2006 and Zimmerman 2009), and sensitivity of estimates to different conceptions of perceived execution risk (Mocan and Gittings 2003).³⁶ More generally, the panel-data literature suffers from the threat of policy endogeneity, failure to include accurate controls, and a lack of knowledge regarding how potential offenders perceive execution risk. Finally, as noted by Berk (2005) and Donohue and Wolfers (2005), results are highly sensitive to the inclusion of certain states and even certain influential data points (i.e., Texas in 1997). The most careful paper to date is that of Kovandzic, Vieraitis, and Paquette-Boots (2009), who use a dataset spanning a longer period of time, employ an expanded set of control variables, and explore a wide variety of operationalizations of the effect of capital punishment and execution risk. The authors find no evidence of a deterrent effect.

4.3 Sanction Nonlinearities

An additional literature that seeks to estimate the magnitude of deterrence effects does so by exploiting nonlinearities in the severity of sanctions faced by certain offenders. Typically, these studies estimate the incidence of arrest rates for young offenders who are either just below or just above the age of criminal majority—generally either seventeen or eighteen years of age, depending

³⁵The debate continues with recent responses to critiques by Donohue and Wolfers (2005) offered by Zimmerman (2009), Dezhbakhsh and Rubin (2011), and Mocan and Gittings (2010).

³⁶While Mocan and Gittings (2010) provide an extensive summary of the robustness of results reported in Mocan and Gittings (2003), Chalfin, Haviland, and Raphael (2013) point out that the responsiveness of murder to execution risk relies on the assumption that individuals are executed fairly soon (within six years) of a conviction.

on the state. While offenders below a given state's age cutoff are adjudicated as juveniles and face relatively low sanctions risk, offenders who are just above the age of majority are adjudicated as adults and are subject to considerably more severe sanctions. Given that the conditional probability of an arrest is smooth as a function of age around the age of criminal majority, any behavioral response of offenders to the threshold is assumed to be the result of deterrence.

The canonical paper in this literature is that of Lee and McCrary (forthcoming). Using data from Florida, Lee and McCrary document a sizable discontinuity in the probability that a young offender is sentenced to prison depending upon whether the arrest occurred prior to or after the offender's eighteenth birthday. Despite the fact that the expected sentence length for an adult arrestee is over twice as great as that faced by a juvenile offender, Lee and McCrary find little evidence of deterrence. Their estimates suggest an elasticity of crime with respect to sentence lengths of approximately -0.05, an estimate that is far smaller than that of Drago, Galbiati, and Vertova (2009), who estimated an elasticity for Italian adults. Findings in Lee and McCrary are perhaps surprising, but are supported by results reported in Hjalmarsson (2009b), who documents that perceived increases in the severity of sanctions at the age of criminal majority among juvenile offenders are smaller than the actual changes, thus suggesting a mechanism underlying these small effects. The implication is that deterrence is not operational because perceptions do not match the incentives created by public policy.

Lee and McCrary's research design has now been replicated to varying degrees. Most recently, Hansen and Waddell (2014) study the effect of Oregon's age of majority on juvenile offending and report some evidence of a decline in crime upon reaching the age of majority for covered crimes. However, results that utilize an appropriately small bandwidth are not significant at conventional levels indicating, at best, weak evidence in favor of deterrence effects. Finally, in a reduced-form analysis using nationallevel data in the NLSY, Hjalmarsson (2009b) finds little evidence of deterrence around state-specific ages of majority using self-reported data on offending.

A final paper worth mentioning is that of Hjalmarsson (2009b), which studies the effect of serving time in prison on subsequent arrest among juvenile offenders in Washington State. Exploiting a discontinuity in the state's sentencing guidelines, Hjalmarsson reports that incarcerated juveniles have lower propensities to be reconvicted of a crime. This deterrent effect is also observed for older and more criminally experienced offenders. The differential findings in Hjalmarsson (2008), on the one hand, and Hjalmarsson (2009b) and Lee and McCrary (forthcoming), on the other hand can potentially be rationalized by the fact that while the latter studies considered the behavioral response to a general threat of punishment, the former study measures the behavioral response to actual punishment that has already been experienced.

On the whole, the RD literature around the age of criminal majority produces little evidence of deterrence among young offenders. The available evidence suggests that this may, in part, be due to a lack of awareness of the size of the sanctions discontinuity, leaving open the possibility that deterrence may be found if the discontinuity is "advertised" as in pulling-levers-type focused-deterrence strategies. A remaining issue concerns the focus of the literature on arrests, which are an imperfect proxy for offending. In particular, if police officers are less likely to arrest an individual just below the age of majority relative to an individual just above the age of majority for a given crime, the resulting RD estimates will be attenuated with respect

to the actual change in offending, which may well have been positive. While no direct evidence suggests that this type of officer behavior is widely employed, the concern is worth noting.

4.4 Deterrence versus Incapacitation

As with the literature examining the response of crime to the certainty of apprehension, the primary conceptual challenge to interpreting the empirical literature on sanctions is that it is difficult to discern between deterrence and incapacitation. With respect to studies of the crime-prison population elasticity, two issues merit discussion. First, the size of a state's prison population is only a proxy for the punitiveness of the sanctions regime. In practice, the size of the prison population is a function of many things: the underlying rate of offending, the certainty of punishment (in part due to the probability of apprehension), and the criminal propensity of the marginal offender when the prison population changes. Prison population is a stock, not a flow, and accordingly when the prison population declines it can be due to either an increase in the contemporary probability of a custodial sentence or to flows out of prison (Durlauf and Nagin 2011). Likewise, deterrence is only one of the mechanisms by which prisons affect crime, the other being incapacitation. For these reasons, the literature that examines the crime–prison population elasticity, while important with respect to public policy, is not particularly informative with respect to deterrence.

In our view, research that studies the instantaneous impact of shocks to the sanctions regime are considerably more informative. Indeed, identifying the sensitivity of crime to a shock to the sanctions regime is conceptually close to testing Becker's prediction that behavior will respond to the severity of a sanction. However, even with perfect identification, attributing a change in offending that occurs in the aftermath of a sanctions shock to deterrence requires a logical leap. In particular, the logical leap is greatest when the sanctions regime becomes more punitive along both the intensive and extensive margin. To the extent that a custodial sentence becomes both longer and more likely, tougher sentencing generates both deterrence and incapacitation effects. This is an issue in interpreting much of the literature on sentence enhancements. Such a concern is addressed in Kessler and Levitt (1999), which studies the effect of California Proposition 8, a 1982 ballot amendment that enhanced the length of sentences for certain felonies, but not for others. Because prior to Proposition 8, each of the felonies already required mandatory prison time, any instantaneous response of crime to Proposition 8 would have to be attributable to deterrence. Kessler and Levitt find that crimes that were eligible for the enhancement fell by between 4 and 8 percent in the aftermath of Proposition 8, relative to a control group of crimes not eligible for the enhancement. The implication of these findings is that increased sanctions promote substantial deterrence. However, while the logic is, in general, persuasive, the validity of Kessler and Levitt's results have been called into question by Webster, Doob, and Zimring (2006), who argue that overall crime did not fall in the aftermath of Proposition 8, and by Raphael (2006), who argues that crimes ineligible for sentence enhancements do not form an appropriate control group for crimes eligible for the enhancement.

Changes in a state's use of capital punishment, in theory, offers a more appropriate means of identifying deterrence. This is because capital murder is sufficiently serious as to warrant a long prison sentence regardless of the specifics of a state's sentencing regime. Hence, when an offender is sentenced to death (as opposed to a sentence of life without the possibility of parole), there is no instantaneous incapacitation effect. With respect to capital punishment, the evidence of deterrence is, at best, mixed with the most rigorous studies failing to find evidence of deterrence. Moreover, the identification problems in the literature are considerable, as it is difficult to identify a shock to a state's capital-punishment regime that is plausibly exogenous. Overall, we do not believe this literature offers any credible evidence of deterrence, though it is not clear that variation in capital-punishment regimes will ever be sufficiently random and that murder rates will ever be sufficiently dense to allow us to credibly detect a treatment effect.

Undoubtedly the best tests for deterrence may be found in research that follows individual offenders who, upon being apprehended, face different sanctions for a given crime. To the extent that differential treatment is driven by arbitrary distinctions within the criminal-justice system, research can identify deterrence by comparing the behavior of offenders who are otherwise similar but are treated differently. Such a research design is truly quasi-experimental in the sense that treatment effects can be interpreted using the language of the Rubin causal model. Moreover, individual-level studies track the behavior of individuals who are not in prison and accordingly are not incapacitated. Hence, any behavioral shift is plausibly attributable to deterrence. These individual-level studies produce mixed evidence. On the one hand, studies of three-strikes laws establish that offenders with two strikes are less likely to reoffend than offenders with one strike (Zimring, Hawkins, and Kamin 2001 and Helland and Tabarrok 2007). Likewise, in Drago, Galbiati, and Vertova's study of Italy's clemency bill, prisoners who faced harsher sanctions upon being rearrested were less likely to be rearrested. On the other hand, studies of sanction nonlinearities in which offenders of slightly different ages receive differential treatment report little evidence of a large deterrence effect. Of course, these

results might be rationalized by differences in the responsiveness to a sanction among offenders of different ages.

To date, the degree to which offenders are deterred by harsher sanctions remains an open question. Undoubtedly, deterrence can exist in extreme circumstances in which the punishment is immediate and harsh. Likewise, evidence of deterrence is found when punishment severity faced by individual offenders is both extraordinarily severe and known. However, within the range of typical changes to sanctions in contemporary criminal-justice systems, the evidence suggests that the magnitude of deterrence owing to more severe sentencing is not large and is likely to be smaller than the magnitude of deterrence induced by changes in the certainty of capture. What is less well understood is the extent to which changing sentencing severity along the extensive margin induces deterrence. Since this increases the severity of punishment in the near rather than the distant future, one might think that deterrence effects will be more easily observed.

5. Work and Crime

The final pillar of the neoclassical model of crime considers the responsiveness of crime to a carrot (better employment opportunities) rather than a stick (certainty or severity of punishment). In particular, since the benefit of a criminal act must be weighed against the value of the offender's time spent in an alternative activity, an increase in the opportunity cost of an offender's time can be thought of as a deterrent to crime. Indeed, this principle has generated considerable public support for a variety of policies designed to reduce recidivism among offenders returning from prison—for example, the provision of job training, employment counseling, and transitional jobs.

The empirical literature examining the impact of local labor-market conditions on

crime can be divided into two related but distinct research literatures. The first literature examines the relationship between unemployment and crime. A second literature examines the impact of the responsiveness of crime to wages.³⁷ With respect to both literatures, approaches to study the effect of labor markets on crime are varied and include papers that use individual microdata, as well as state- or county-level variation. Taken as a whole, the literature that uses aggregate data to disentangle the effect of economic conditions on crime presents a mixed picture. In general, results are sensitive to the time period studied, the population under consideration, the type of wage or unemployment rate that is employed, as well as the criminal offenses analyzed. However, more recent and carefully identified papers tend to find evidence of a fairly robust relationship between both unemployment and wages and crime. There is also a literature that examines the relationship between crime, unemployment, and wages using individual data. We discuss the implications of this literature for the study of deterrence in the final part of this section.

5.1 Unemployment

Periods of unemployment are thought to generate incentives to engage in criminal activity, either as a means of income supplementation or consumption smoothing or, more generally, due to the effect of psychological strain (Chalfin and Raphael 2011). To the extent that a decline in unemployment raises the opportunity cost of crime without generating a subsequent increase in the probability of apprehension or the severity of the expected sanction, the response of crime to changes in the unemployment rate can be thought of as capturing, among a host of behavioral responses, deterrence.

In general, the early literature linking unemployment and crime has produced mixed and frequently contradictory results, leading Chiricos (1987) to characterize scholarly opinion on the topic as a "consensus of doubt." In particular, Chiricos found that, among the studies he reviewed, fewer than half found significant positive effects of aggregate unemployment rates on crime rates.³⁸ This conclusion is echoed in reviews by Freeman (1983), Piehl (1998), Mustard (2010), and Chalfin and Raphael (2011).

Recent literature on the topic of unemployment and crime has benefited from several methodological advances-in particular, the use of panel data as opposed to a cross-sectional data or national time series. Examples of panel-data research include Entorf and Spengler (2000) for Germany, Papps and Winkelmann (2000) for New Zealand, Machin and Meghir (2004) for the United Kingdom, Andresen (2013) for Canada, and Arvanites and Defina (2006), Ihlanfeldt (2007), Rosenfeld and Fornango (2007), and Phillips and Land (2012) for the United States. With the exception of Papps and Winkelmann (2000), each of these papers finds at least some evidence in favor of a link between unemployment and crime, in particular, property crime.

³⁷There is also a large and growing experimental literature that evaluates how at-risk individuals have responded to the provision of job coaching, employment counseling, career placement, and other employment-based services.

³⁸Nonetheless, Chiricos's review also found that the unemployment–crime relationship was three times more likely to be positive than negative and fifteen times more likely to be positive and significant than negative and significant, indicating a basis for further research. The results were especially strong for property crimes—in particular, larceny and burglary. Chiricos suggests that research results are generally consistent by level of aggregation, though they tend to be more consistently positive and significant at lower levels of aggregation. This hypothesis is echoed by Levitt (2001), who likewise argues that nationallevel time-series analyses obscure the unemployment– crime relationship by failing to account for rich variation across space.

A second innovation in the recent literature has been to employ instrumental variables to address the potential endogeneity between labor-market conditions and crime. The first such study is that of Raphael and Winter-Ebmer (2001), who use a state-level panel data set covering 1979-98 to study the effect of unemployment rates on various types of crime employing two instruments for the unemployment rate-the value of military contracts with the federal government, as well as the regional impact of shocks to the price of oil. For property-crime rates, the results consistently indicate a positive effect of unemployment on crime with a 1 percentage point increase in the unemployment rate predicting a 3-5 percent increase in property crime. For violent crime, however, the results are mixed. Gould, Weinberg, and Mustard (2002) provide a similar analysis at the county level, using a county's initial industry mix and measures of skill-biased technical change as an instrument for unemployment. They too find evidence of a positive relationship between unemployment and crime, particularly property crime. Taken as a whole, results reported in Raphael and Winter-Ebmer (2001) and Gould, Weinberg, and Mustard (2002) imply that variation in unemployment rates explained between 12 percent and 40 percent of the decline in property crime during parts of the 1990s. In a more recent paper, Lin (2008) builds on these approaches using exchange-rate shocks to isolate exogenous variation in unemployment rates. Lin reports that a 1 percentage point increase in unemployment leads to a 4 to 6 percent decline in property crime and would explain roughly one-third of the crime drop during the 1990s.³⁹

On the whole, the preponderance of the evidence suggests that there is an important

relationship between unemployment rates and property crime, but little impact of unemployment on violent crime, a conclusion echoed in a recent review by Cook (2010). In the recent literature, which is more careful with respect to addressing omitted variables bias and simultaneity, the relationship between unemployment and property crime is found regardless of the level of aggregation (counties or states).⁴⁰ The relationship between unemployment and property crime is empirically meaningful, as property crime would be predicted to rise by between 9 and 18 percent during a serious recession in which unemployment increased by 3 percentage points. Moreover, this, if anything, may understate the magnitude of the relationship, as crime appears to be particularly sensitive to the existence of employment opportunities for low-skilled men (Schnepel 2013). Nevertheless, the estimates remain sensitive to the time period studied. To wit, property crime has generally continued to decline through the recent Great Recession, which increased unemployment rates nationally by as many as 4 percentage points.

5.2 Wages

A second and related research literature considers the impact of wage levels on crime rates. There are several a priori reasons to expect a stronger relationship between

³⁹Fougere, Kramarz, and Pouget (2009) provide a similar analysis for France, finding effects that are similar in magnitude.

⁴⁰Prominent IV papers, including Raphael and Winter-Ebmer (2001); Gould, Weinberg, and Mustard (2002); and Lin (2008) do not uniformly find that instrumenting results in a more positive relationship between unemployment and crime as would be predicted by the omission of procyclical control variables or simultaneity bias. Another explanation for slippage between least squares and IV estimates of the effect of unemployment rate. To the extent that such errors are classical, attenuation bias will mean that 2SLS estimates will exceed ordinary least squares (OLS) estimates. To the extent that this pattern is not found, there is the possibility that OLS estimates are actually upward biased due to simultaneity or omitted variables.

crime and wages than between crime and unemployment. First, as noted by Gould, Weinberg, and Mustard (2002), since criminal participation is associated with a set of fixed costs, crime may well be more responsive to long-term labor-market measures such as levels of human capital or wages than unemployment spells, which are typically ephemeral. Second, at any given time, the number of individuals who are employed in low-wage jobs vastly outnumbers the number of unemployed and, as such, wages for unskilled men may play a proportionally greater role than unemployment in encouraging crime (Hansen and Machin 2002). In fact, among individuals who reported engaging in crime during the past year, a large majority reported wage earnings (Grogger 1998) and three-quarters were employed at the time of their arrest, indicating that the behavior of a majority of offenders should be sensitive to changes in the wage.

The literature linking wages to crime has, in general, generated more consensus than the unemployment literature. Prominent panel-data papers include Doyle, Ahmed, and Horn (1999), who analyze state-level panels for 1984–93 and find that higher average wages reduce both property and violent crime (elasticity estimates vary between -0.3 and -0.9), and Gould, Weinberg, and Mustard (2002), who restrict their analysis to the wages of relatively low-skilled men and find, using a county-level panel spanning 1979 to 1997, that the falling wages of unskilled men in this period led to an 18 percent increase in robbery, a 14 percent increase in burglary, and a 7 percent increase in larceny. These findings are striking in that they indicate that wage trends explain more than half of the increase in both violent and property crimes over the entire period.

In a similar analysis for the United Kingdom, Machin and Meghir (2004) examine changes in regional crime rates in relation to changes in the tenth and twenty-fifth percentiles of the region's wage distribution and focus on the retail sector, an industry where low-skilled workers have the ability to manipulate their hours of work. They find that crime rates are higher in areas where the bottom of the wage distribution is low. With regard to microdata, Grogger (1998), leveraging data from the NLSY, finds that youth wages account for approximately three-quarters of the variation in youth crime. Finally, a related literature considers the responsiveness of crime to minimum wages and consistently finds evidence in favor of a negative relationship between the two variables (Corman and Mocan 2005 and Fernandez, Holman, and Pepper 2014).

5.3 Individual-Level Studies

In addition to aggregate-level studies that examine the impact of macroeconomic fluctuations on crime, there is a parallel literature that studies the effect of wage shifts and job loss—mainly job loss—over the life course. This literature uses natural variation to examine whether offending rises when individuals find themselves out of work. An early example of such research is that of Crutchfield and Pitchford (1997), who, using data from the 1979 NLSY79, find that the approximately 8,000 adults who responded to the first wave of the survey were more likely to engage in crime when they are out of the labor force and when they expect their current job to be of short duration. A host of similar cohort studies have found similar correlations among males in London, as well as individuals born in Philadelphia in 1945 (Thornberry and Christenson 1984 and Witte and Tauchen 1993). For several reasons, we do not believe this literature has great value in uncovering deterrence effects. First, even conditioning on fixed effects, individual-level models do not plausibly account for omitted variation that may be related to both unemployment and offending. In particular, a change in an

important unobserved factor may drive both spells out of work and criminal activity. For example, illegal drug use may simultaneously cause both an unemployment spell as well as participation in crime. Alternatively, other life stresses, problems with personal relationships, mental health problems, etc. may cause the simultaneous co-occurrence of unemployment (or underemployment) and criminal activity (Chalfin and Raphael 2011). To be sure, such issues of causal identification pose a challenge in all micro-level social science research using observational data. Nonetheless, absent a clear source of exogenous variation in employment status or employment prospects, one should probably consider these sorts of longitudinal estimates as providing an upper bound on the likely effect sizes.

A second individual-level literature is, in our view, more useful. This literature considers the impact of providing employment services, or, in some cases, transitional jobs to former prisoners—in particular assessing whether such programs reduce recidivism. The research is predominantly comprised of randomized experiments and is, as such, highly credible. Experimental interventions of this nature tend to include programs that provide income, employment-based services, or skills-building social services. There are over a dozen experimental evaluations of such efforts in the United States, in which treatment group members are randomly assigned. A key advantage of these studies is that the treatment is clearly exogenous, and, as such, any observed impacts plausibly represent true causal effects. However, the reader should be careful in interpreting the results of these programmatic interventions, as it is often the case that many members of the randomized control group receive similar services elsewhere. Often, it is difficult to document such contamination and it is not always self-evident that the intervention has a large marginal impact on service

delivery. Second, since these interventions are targeted at particular groups with offense histories that cross fairly stringent severity levels (former prisoners, for example), they tend to treat individuals who may not be particularly responsive to positive incentives.

Community-based employment interventions became popular in the United States beginning in the 1970s. Under authority of the 1962 Manpower Development and Training Act, the US Department of Labor launched a number of programs aimed at former prisoners beginning with the Living Insurance for Ex-Prisoners program, which provided a living stipend and job-placement assistance to prisoners returning to Baltimore between 1972 and 1974, and the Transitional Aid Research Project (TARP), which provided various combinations of cash assistance and job-placement services to five different experimental groups of ex-offenders in Georgia and Texas. The earlier demonstration evaluated by Mallar and Thorton (1978) found significant impacts of the income-support program, with considerably lower offending rates among the treatment group. However, the evaluation of the larger-scale TARP program (Rossi, Berk, and Lenihan 1980) found little effect. The latter evaluation also found a large negative effect of the transitional cash assistance on the labor supply of released inmates. In fact, the authors speculate that the lack of an overall impact on recidivism reflected the offsetting effects of the reduction in recidivism due to the cash assistance and the increased criminal activity associated with being idle (Rossi, Berk, and Lenihan 1980).

There have also been several high-quality evaluations of the impact of providing transitional employment to former inmates. The National Supported Work (NSW) program (recently reanalyzed by Uggen 2000) and the New York Center for Employment Opportunities currently under evaluation by the Manpower Development Research Corporation (MDRC) (Bloom et al. 2007) find some evidence that providing prison releases with transitional employment forestalls recidivism during the two years post-release. However, these programs found considerable heterogeneity in program impacts with the NSW, finding significant effects for older releases and the CEO evaluation reporting significant effects for only those most recently released from prison. Moreover, the majority of the literature reports little evidence of an effect of employment status on recidivism among reentering prisoners (Visher, Winterfield, and Coggeshall 2005). It may well be that employment deters crime among individuals without prison experience, a mechanism that plausibly underlies relationships between measures of macroeconomic performance and crime. However, among individuals with serious criminal records, the evidence of deterrence is difficult to find.

5.4 Identifying Deterrence

A first-order issue in interpreting research on the effect of police and prisons on crime concerns the extent to which deterrence can be disentangled from incapacitation. This issue is not relevant in considering the responsiveness of crime to changes in wages or employment conditions. Nevertheless, it is worth considering whether a significant coefficient on the wage or unemployment rate in a crime regression necessarily identifies deterrence. In particular, for several reasons, crime and unemployment may be related due to factors other than deterrence. First, a relationship between macroeconomic conditions and crime may exist due to the relationship between macroeconomic conditions and criminal opportunities (Cook and Zarkin 1985). For example, during a recession, auto thefts tend to decline presumably because fewer people are employed and therefore drive their cars less frequently (Cook 2010). Second, employment conditions and crime may be linked through behavioral changes that cannot be properly characterized as deterrence. For example, a displaced worker may well develop feelings of anger or loss of control that subsequently manifest in violent behavior. In such a case, the job may not be protective against crime through any deterrence mechanism per se. Nevertheless, a robust relationship between economic conditions and crime is potentially consistent with the idea that individuals respond to incentives, at least at the margin.

6. Conclusion

We reviewed three large literatures regarding the responsiveness of crime to police, sanctions, and local labor-market opportunities. Three key conclusions are worth noting. First, there is robust evidence that crime responds to increases in police manpower and to many varieties of police redeployments. With respect to manpower, our best guess is that the elasticity of violent crime and property crime with respect to police are approximately -0.4 and -0.2, respectively. The degree to which these effects can be attributed to deterrence as opposed to incapacitation remains an open question, though analyses of arrest rates suggests a role for deterrence (Levitt 1998 and Owens 2013). With respect to deployments, experimental research on hot-spots policing and focused deterrence efforts have, in some cases, led to remarkably large decreases in offending, a fact that may be attributable to the visibility of such policies.

Second, while the evidence in favor of a crime–sanction link generally favors relatively small deterrence effects, there does appear to be some evidence of more meaningful deterrence induced by policies that target specific offenders with sentence enhancements. This is seen in the effect of California's three-strikes law on the behavior of offenders with two strikes (see Helland and Tabarrok 2007) and in the behavior of pardoned Italian offenders (Drago, Galbiati, and Vertova 2009). On the other hand, while the elasticity of crime with respect to sentence lengths appears to be large in the Italian case, it is quite small in the California case.

Finally, there is fairly strong evidence, in general, of a link between local labor-market conditions, proxied using the unemployment rate or the wage, on crime. While these effects are unlikely to be appreciably contaminated by incapacitation effects, they may reflect behavioral responses aside from deterrence. Moreover, it is not clear that supplying employment deters offending among the most criminally productive individuals.

Overall, the evidence suggests that individuals respond to the incentives that are the most immediate and salient. While police and local labor-market conditions influence costs that are borne immediately, the cost of a prison sentence, if experienced at all, is experienced sometime in the future. To the extent that offenders are myopic or have a high discount rate, deterrence effects will be less likely. Moreover, given that an empirical finding of deterrence depends on the existence of perceptual deterrence, it may be the case that potential offenders are more aware of changes in policing and local labormarket conditions than they are of changes in incarceration policy, with the exception of specific sentence enhancements that are individually targeted. In the final section of this article, we return to Becker's economic model of crime in an attempt to rationalize the empirical literature with the theory that precipitated it. We close with a couple of concrete recommendations for future work.

6.1 Rationalizing Theory and Empirics

A natural starting place in reconciling the empirics with the theory is to consider that Becker's model does not explicitly predict that offending will be more responsive to

changes in p as opposed to f. This is because while the model allows for varying degrees of risk aversion, there is no other mechanism within the model that can lead to a differential response of crime to p and f that we tend to observe in the empirical literature. However, using more recent theoretical work that builds upon Becker as a guide, we note that there are at least three strong theoretically motivated reasons to expect that crime will be more responsive to p than to f. This is not to say that the Becker model is incorrect—on the contrary, our reading of the empirical literature is that it provides support for the model. Instead, we prefer to characterize the model as a useful starting point for thinking about some of the more nuanced aspects of deterrence.

Perhaps the most enduring criticism of the original neoclassical model is that it is static and, accordingly, does not explicitly allow for individuals to differ in their time preferences, a fact that is inspired by a generation of behavioral economics research and was noted in early work by Block and Heineke (1975) and Cook (1980). Dynamic extensions of Becker can be found in Polinsky and Shavell (1999), Lee and McCrary (forthcoming), and McCrary (2011), among others. An abbreviated version of such a dynamic model was presented in section 1 of this article. The most important insight arising from the dynamic corollary of Becker is that individuals who are myopic and engage in hyperbolic discounting will be more strongly deterred by changes in p, which affect utility immediately, than by changes in *f*, which, for the most part, affect utility in the distant future. To the extent that offenders tend to be hyperbolic discounters and there is ample evidence that many are, we should not expect long sentences to deter to nearly the same degree as changes in the probability of either arrest or some type of punishment.

In recent years, the Becker model has also been augmented to allow for the fact

that sanctions do not necessarily follow from apprehension. Indeed, many offenders are arrested but do not suffer anything more than a trivial criminal sanction, either because charges are not filed or are dismissed or because a custodial sentence is not handed down or is considered already served at the time of sentencing. Likewise, behavioral scientists have incorporated the notion that individuals suffer both legal and nonlegal sanctions when they are arrested for committing a crime (Nagin and Paternoster 1994, Nagin and Pogarsky 2003, and Williams and Hawkins 1986). While the effect of legal sanctions has been well understood since Becker, the extent to which individuals suffer social stigma and other social costs as a result of an arrest remains less well understood. Nagin (2013) formalizes these related concepts within the framework of the Becker model, conceptualizing p as the product of the probability of apprehension, P_a , and the conditional probability of a sanction given apprehension, $P_{(S|a)}$. With these additions, it can be shown that if individuals are more sensitive to changes in the probability of apprehension than to changes in the sanction, it is easy to see that changes in utility (and thus crime) can potentially be explained by informal sanction costs alone (Nagin 2013). A related and arguably more important insight is that in the event that informal sanctions costs are very high, it is considerably more likely that deterrence will accrue via the probability of apprehension (p_a) than via f.

Related to this is the issue of heterogeneity in individual utility over the legal sanction. In the event that the stigma of a custodial sentence declines, the disutility derived from punishment will fall as a function of the length of an individual's criminal history (Durlauf and Nagin 2011 and Nagin, Cullen, and Jonson 2009). That is, as individuals accumulate a longer criminal record, the stigma from being labeled a "criminal" may lose its effectiveness and thus no longer represent an important component of the cost of committing a crime. Likewise, the disutility of punishment may decline due to the presence of informal social networks that develop as a result of an individual's prison experience. Hence, to the extent that a large proportion of crime is committed by repeat offenders, crime should be more sensitive to the probability of apprehension than to the severity of sanction, because repeat offenders have already paid the informal costs associated with being labeled a criminal.

A final extension that merits discussion can be found in Durlauf and Nagin (2011) who consider that, in addition to exhibiting strong time preferences, individuals may have a great deal of trouble accurately estimating the risk of apprehension even after updating in response to new information (Anwar and Loughran 2011). Durlauf and Nagin develop a model in which p is probabilistic and show that for a fixed sanction, when pis perceived to be arbitrarily close to either zero or one (i.e., the probability of apprehension is thought to be either extremely low or extremely high), the effect of certainty will be greatest. This follows from the tendency of individuals to systematically overestimate the likelihood of rare events (e.g., a terrorist attack) and underestimate the likelihood of more common events (e.g., a car accident). As Durlauf and Nagin note, in a world in which the perceived probability of detection is very low, even small changes in that perceived probability can have correspondingly large effects, thus potentially rationalizing large behavioral responses to hot-spots policing and deterrence advertising.

The implication that the response of crime to p might be systematically greater than to f has far-reaching implications with respect to public policy. First, given that deterrence is cheap relative to incapacitation, efficient resource allocation demands a shift in resources toward more deterrence-intensive

inputs. Hence, in deciding how to allocate criminal-justice resources between police officers and prisons, the available evidence suggests that money is best spent on police officers, as well as perhaps on jails that might be used to detain individuals who receive short sentences. Second, it may be possible to reduce the size of the US prison population, which has grown six-fold over the course of a generation, without compromising public safety. Such an idea is strongly supported by the small crime-prison population elasticities reported by Liedka, Piehl, and Useem (2006) and Johnson and Raphael (2012), as well as Lofstrom and Raphael's recent evaluation of the effects of California's realignment policy. It is also articulated anecdotally by the experience of New York State, which has both reduced its prison population as well as its crime rate in recent years.

These points are underscored in Kleiman (2009) and Durlauf and Nagin (2011), among others, who ask the provocative question as to whether both imprisonment and crime can be simultaneously reduced. The answer to this question lies in the magnitude of the relevant deterrence elasticities, the specifics of which are cleverly illustrated in Durlauf and Nagin, who ask us to consider a world in which there are two types of criminal opportunities: desirable opportunities with corresponding probability of arrest, p_0 , and undesirable opportunities with probability of arrest, p_1 . Initially, individuals who encounter a criminal opportunity that is assigned probabilistically elect to commit a crime if $p > p^*$, where $p^* > p_1 > p_0$. Hence, all available opportunities are acted upon. Durlauf and Nagin next ask us to consider that p_0 and p_1 are each increased by a factor g such that $gp_0 < p^* < gp_1$. In this world, only the desirable opportunities will be acted upon. The obvious result is that crime declines as all of the opportunities are no longer attractive. What is less obvious is that clearance rates fall by construction and, accordingly, so do the number of arrests and subsequent incarcerations. As shown in Blumstein and Nagin (1978), if the magnitude of either the elasticity of crime with respect to p or f is less than 1, then the decline in the crime rate associated with an increase in p and f will not be sufficiently large to avoid an increase in the incarceration rate. The key then is to determine whether there are policies that satisfy this inequality or, in the absence of such policies, identify policies that decrease severity but have large elasticities with respect to p. In our view, swift-and-certain sanctions regimes such as that motivated by HOPE and visible police presence are two such policies that seem especially promising.

6.2 Directions for Future Research

As each of the deterrence literatures has matured, researchers can begin to focus less attention on standard identification problems and more attention on identifying mechanisms. This is not to say that causal identification is unimportant. On the contrary—we believe it is as important as it ever was. However, in the past decade, great strides have been made in selecting strategies that credibly identify the causal effects of a variety of policies which has, in turn, generated a consensus on the range of effects that can be expected to accrue from a given type of intervention. For several literatures that have generated no such consensus (e.g., capital punishment), we strongly suspect that there is little progress to be made in rehashing the same state-level data. Accordingly, we provide three recommendations to guide future work.

First, there is a large and growing literature that supports the deterrence value of hot-spots policing and, in particular, the importance of a visible police presence. However, the evidence on the effects of disorder policing is more mixed and the idea remains highly controversial. The best evidence suggests that cleaning up physical disorder is important, but it is not clear whether broken-windows policing is a necessary ingredient to this strategy. Given the inherent risks of broken-windows policing that accrue to both officers and citizens and the civil-rights concerns that are intrinsic in such a strategy, future research is needed to identify the extent to which broken windows reduces crime and, if so, whether those reductions are due to deterrence or incapacitation. There continues to be substantial disagreement on this topic and, while we are skeptical of the deterrence value of broken windows, our reading of the research literature is that we do not yet have sufficient information to make an informed recommendation.

Second, while evidence is building that swift-and-certain sanctions can deter offending at dramatically lower costs for both society and offenders, the idea requires additional testing. In particular, the conditions under which such programs work and the degree to which they are replicable and scalable remains unknown. It has been suggested that the success of such programs often depends on the existence of effective leaders and an unusual degree of cooperation among local policy makers. Moreover, swift-and-certain sanctioning only works if offending can be reliably detected in the first place.

A third area that, in our view, will benefit from greater research, concerns the deterrence value of investments in private precautions by potential victims. Evaluations of LoJack (Ayres and Levitt 1998) and business improvement districts (Cook and MacDonald 2011) establish that private investments can deter (after all, they cannot incapacitate). However, we know relatively little about the effects of other types of private behavior, such as investments in burglary alarm systems, the emergence of various smart phone apps that provide information, as well as other types of technology. In closing, we note that Gary Becker's recent passing prompts us to acknowledge yet again the decisive impact of his landmark 1968 paper. As his ubiquity in this review makes clear, his insights launched an entire literature—one that has had and continues to have profound implications for, and impact on, public policy and safety.

References

- Allingham, Michael G., and Agnar Sandmo. 1972. "Income Tax Evasion: A Theoretical Analysis." *Journal of Public Economics* 1 (3–4): 323–38.
- Andenaes, Johannes. 1974. Punishment and Deterrence. Ann Arbor: University of Michigan Press.
- Andresen, Martin A. 2013. "Unemployment, Business Cycles, Crime, and the Canadian Provinces." *Journal* of Criminal Justice 41 (4): 220–27.
- Anwar, Shamena, and Thomas A. Loughran. 2011. "Testing a Bayesian Learning Theory of Deterrence among Serious Juvenile Offenders." *Criminology* 49 (3): 667–98.
- Apel, Robert. 2013. "Sanctions, Perceptions, and Crime: Implications for Criminal Deterrence." *Jour*nal of Quantitative Criminology 29 (1): 67–101.
- Apel, Robert, and Daniel S. Nagin. 2011. "General Deterrence: A Review of Recent Evidence." In *Crime and Public Policy*, Fourth edition, edited by James Q. Wilson and Joan Petersilia, 411–36. Oxford and New York: Oxford University Press.
- Arvanites, Thomas M., and Robert H. Defina. 2006. "Business Cycles and Street Crime." *Criminology* 44 (1): 139–64.
- Ayres, Ian, and Steven D. Levitt. 1998. "Measuring Positive Externalities from Unobservable Victim Precaution: An Empirical Analysis of Lojack." *Quarterly Journal of Economics* 113 (1): 43–77.
- Baldry, Jonathan. 1974. "Positive Economic Analysis of Criminal Behavior." In *Economic Policies and Social Goals: Aspects of Public Choice*, edited by A. J. Culyer, 171–98. London: Martin Robertson.
- Beccaria, Cesare. 1764. On Crimes and Punishments. Oxford: Clarendon Press, 1957.
- Becker, Gary S. 1968. "Crime and Punishment: An Economic Approach." *Journal of Political Economy* 76 (2): 169–217.
- Bell, Brian, Laura Jaitman, and Stephen Machin. 2014. "Crime Deterrence: Evidence from the London 2011 Riots." *Economic Journal* 124 (576): 480–506.
- Benson, Bruce L., and Brent D. Mast. 2001. "Privately Produced General Deterrence." *Journal of Law and Economics* 44 (Supplement 2): 725–46.
- Bentham, Jeremy. 1789. An Introduction to the Principles of Morals and Legislation. Oxford: Clarendon Press.
- Berk, Richard. 2005. "New Claims about Executions and General Deterrence: Déjà Vu All Over Again?"

Journal of Empirical Legal Studies 2 (2): 303–30.

- Berk, Richard, and John MacDonald. 2010. "Policing the Homeless: An Evaluation of Efforts to Reduce Homeless-Related Crime." *Criminology and Public Policy* 9 (4): 813–40.
- Block, M. K., and J. M. Heineke. 1975. "A Labor Theoretic Analysis of the Criminal Choice." American Economic Review 65 (3): 314–25.
- Bloom, Dan, Cindy Redcross, Janine Zweig, and Gilda Azurdia. 2007. "Transitional Jobs for Ex-Prisoners: Early Impacts from a Random Assignment Evaluation of the Center for Employment Opportunities Prisoner Reentry Program." MDRC Working Paper.
- Blumstein, Alfred. 1995. "Youth Violence, Guns, and the Illicit-Drug Industry." *Journal of Criminal Law* and Criminology 86 (1): 10–36.
- Blumstein, Alfred, and Daniel Nagin. 1978. "On the Optimum Use of Incarceration for Crime Control." *Operations Research* 26 (3): 381–405.
- Braga, Anthony A. 2001. "The Effects of Hot Spots Policing on Crime." Annals of the American Academy of Political and Social Science 578 (1): 104–25.
- Braga, Anthony A. 2005. "Hot Spots Policing and Crime Prevention: A Systematic Review of Randomized Controlled Trials." *Journal of Experimental Criminology* 1 (3): 317–42.
- Braga, Anthony A. 2008a. "Police Enforcement Strategies to Prevent Crime in Hot Spot Areas." Washington, DC: US Department of Justice Office of Community Oriented Policing Services.
- Braga, Anthony A. 2008b. Problem-Oriented Policing and Crime Prevention, Second edition. Monsey, NY: Willow Tree Press.
- Braga, Anthony A. 2008c. "Pulling Levers Focused Deterrence Strategies and the Prevention of Gun Homicide." *Journal of Criminal Justice* 36 (4): 332–43.
- Braga, Anthony A., and Brenda J. Bond. 2008. "Policing Crime and Disorder Hot Spots: A Randomized Controlled Trial." Criminology 46 (3): 577–607.
- Braga, Anthony A., David M. Kennedy, Elin J. Waring, and Anne Morrison Piehl. 2001. "Problem-Oriented Policing, Deterrence, and Youth Violence: An Evaluation of Boston's Operation Ceasefire." *Journal of Research in Crime and Delinquency* 38 (3): 195–225.
- Braga, Anthony A., and David I. Weisburd. 2012. "The Effects of 'Pulling Levers' Focused Deterrence Strategies on Crime." *Campbell Systematic Reviews* 6.
- Braga, Anthony A., David L. Weisburd, Elin J. Waring, Lorraine Green Mazerolle, William Spelman, and Francis Gajewski. 1999. "Problem-Oriented Policing in Violent Crime Places: A Randomized Controlled Experiment." Criminology 37 (3): 541–80.
- Bratton, William J., and George L. Kelling. 2015. "Why We Need Broken Windows Policing." *City Journal* Winter.
- Britt, Chester L. 1997. "Reconsidering the Unemployment and Crime Relationship: Variation By Age Group and Historical Period." *Journal of Quantitative Criminology* 13 (4): 405–28.
- Brown, William W., and Morgan O. Reynolds. 1973.

"Crime and 'Punishment': Risk Implications." *Journal of Economic Theory* 6 (5): 508–14.

- Buonanno, Paolo, and Steven Raphael. 2013. "Incarceration and Incapacitation: Evidence from the 2006 Italian Collective Pardon." *American Economic Review* 103 (6): 2437–65.
- Burdett, Kenneth, Ricardo Lagos, and Randall Wright. 2004. "An On-the-Job Search Model of Crime, Inequality, and Unemployment." *International Economic Review* 45 (3): 681–706. Caetano, Gregorio, and Vikram Maheshri. 2014. "Identifying Dynamic Spillovers in Criminal Behavior." Unpublished.
- Caeti, Tory J. 1999. "Houston's Targeted Beat Program: A Quasi-experimental Test of Police Patrol Strategies." Ph.D. Dissertation, Sam Houston State University.
- Cameron, Samuel. 1988. "The Economics of Crime Deterrence: A Survey of Theory and Evidence." Kyklos 41 (2): 301–23.
- Carmichael, Stephanie E., and Alex R. Piquero. 2006. "Deterrence and Arrest Ratios." *International Journal of Offender Therapy and Comparative Criminology* 50 (1): 71–87.
- Chalfin, Aaron, Amelia M. Haviland, and Steven Raphael. 2013. "What Do Panel Studies Tell Us About a Deterrent Effect of Capital Punishment? A Critique of the Literature." *Journal of Quantitative Criminology* 29 (1): 5–43.
- Chalfin, Aaron, and Justin McCrary. Forthcoming. "Are U.S. Cities Underpoliced? Theory and Evidence." *Review of Economics and Statistics.*
- Chalfin, Aaron, and Steven Raphael. 2011. "Work and Crime." In *The Oxford Handbook of Crime and Criminal Justice*, edited by Michael Tonry, 444–78. Oxford and New York: Oxford University Press.
- Chalfin, Aaron, and Sarah Tahamont. Forthcoming. "The Effects of Prisons on Crime." In *The Oxford Handbook of Prisons and Imprisonment*, edited by John Wooldredge and Paula Smith. Oxford and New York: Oxford University Press.
- Charles, Kerwin Kofi, and Steven N. Durlauf. 2013. "Pitfalls in the Use of Time Series Methods to Study Deterrence and Capital Punishment." *Journal of Quantitative Criminology* 29 (1): 45–66.
- Chen, Daniel L. 2013. "The Deterrent Effect of the Death Penalty? Evidence from British Commutations during World War I." NBER Working Paper.
- Chiricos, Theodore G. 1987. "Rates of Crime and Unemployment: An Analysis of Aggregate Research Evidence." *Social Problems* 34 (2): 187–212.
- Cochran, John K., Mitchell B. Chamlin, and Mark Seth. 1994. "Deterrence or Brutalization? An Impact Assessment of Oklahoma's Return to Capital Punishment?" Criminology 32 (1): 107–34.
- Cohen, Jacqueline, and Jens Ludwig. 2003. "Policing Crime Guns." In *Evaluating Gun Policy: Effects* on Crime and Violence, edited by Jens Ludwig and Philip J. Cook, 217–50. Washington, DC: Brookings Institution Press.
- Cook, Philip J. 1980. "Research in Criminal Deterrence: Laying the Groundwork for the Second

Decade." Crime and Justice 2: 211–68.

- Cook, Philip J. 2010. "Property Crime—Yes; Violence— No: Comment on Lauritsen and Heimer." Criminology and Public Policy 9 (4): 693–97.
- Cook, Philip J., and John MacDonald. 2011. "Public Safety through Private Action: An Economic Assessment of BIDS." *Economic Journal* 121 (552): 445–62.
- Cook, Philip J., and Gary A. Zarkin. 1985. "Crime and the Business Cycle." *Journal of Legal Studies* 14 (1): 115–28.
- Corman, Hope, and Naci Mocan. 2005. "Carrots, Sticks, and Broken Windows." *Journal of Law and Economics* 48 (1): 235–66.
- Corsaro, Nicholas, Rod K. Brunson, and Edmund F. McGarrell. 2010. "Evaluating a Policing Strategy Intended to Disrupt an Illicit Street-Level Drug Market." *Evaluation Review* 34 (6): 513–48.
- Corsaro, Nicholas, Aaron Chalfin, and Edmund McGarrell. 2014. "An Application of Synthetic Controls for Comparative Case Studies: Examining the Impact of the National Project Safe Neighborhoods Strategy on Urban Gun Violence." Working Paper.
- Corsaro, Nicholas, and Edmund McGarrell. 2010. "Reducing Homicide Risk in Indianapolis between 1997 and 2000." *Journal of Urban Health* 87 (5): 851–64.
- Corsaro, Nicholas, Eleazer D. Hunt, Natalie Kroovand Hipple, and Edmund McGarrell. 2012. "The Impact of Drug Market Pulling Levers Policing on Neighborhood Violence: The Impact of Drug Market Pulling Levers Policing on Neighborhood Violence." Criminology and Public Policy 11 (2): 167–99.
- Crutchfield, Robert D., and Susan R. Pitchford. 1997. "Work and Crime: The Effects of Labor Stratification." Social Forces 76 (1): 93–118.
- DeAngelo, Gregory, and Benjamin Hansen. 2014. "Life and Death in the Fast Lane: Police Enforcement and Traffic Fatalities." *American Economic Journal: Economic Policy* 6 (2): 231–57.
- Dezhbakhsh, Hashem, and Paul H. Rubin. 2011. "From the 'Econometrics of Capital Punishment' to the 'Capital Punishment' of Econometrics: On the Use and Abuse of Sensitivity Analysis." Applied Economics 43 (25): 3655–70.
- Dezhbakhsh, Hashem, Paul H. Rubin, and Joanna M. Shepherd. 2003. "Does Capital Punishment Have a Deterrent Effect? New Evidence from Postmoratorium Panel Data." *American Law and Economics Review* 5 (2): 344–76.
- Dezhbakhsh, Hashem, and Joanna M. Shepherd. 2006. "The Deterrent Effect of Capital Punishment: Evidence from a 'Judicial Experiment'." *Economic Inquiry* 44 (3): 512–35.
- Di Tella, Rafael, and Ernesto Schargrodsky. 2004. "Do Police Reduce Crime? Estimates Using the Allocation of Police Forces after a Terrorist Attack." American Economic Review 94 (1): 115–33.
- Donohue, John J. 2009. "Assessing the Relative Benefits of Incarceration: Overall Changes and the Benefits on the Margin." In *Do Prisons Make Us Safer*? *The Benefits and Costs of the Prison Boom*, edited

by Stephen Raphael and Michael A. Stoll, 269–341. New York: Russell Sage Foundation.

- Donohue, John J., Daniel E. Ho, and Patrick Leahy. 2013. "Do Police Reduce Crime? A Reexamination of a Natural Experiment." In *Empirical Legal Analysis: Assessing the Performance of Legal Institutions*, edited by Yun-chien Chang, 125–44. London and New York: Taylor and Francis, Routledge.
- Donohue, John J., and Justin Wolfers. 2005. "Uses and Abuses of Empirical Evidence in the Death Penalty Debate." Stanford Law Review 58 (3): 791–846.
- Donohue, John J., and Justin Wolfers. 2009. "Estimating the Impact of the Death Penalty on Murder." *American Law and Economics Review* 11 (2): 249–309.
- Doyle, Joanne M., Ehsan Ahmed, and Robert N. Horn. 1999. "The Effects of Labor Markets and Income Inequality on Crime: Evidence from Panel Data." *Southern Economic Journal* 65 (4): 717–38.
- Draca, Mirko, Stephen Machin, and Robert Witt. 2011. "Panic on the Streets of London: Police, Crime, and the July 2005 Terror Attacks." *American Economic Review* 101 (5): 2157–81.
- Drago, Francesco, Roberto Galbiati, and Pietro Vertova. 2009. "The Deterrent Effects of Prison: Evidence from a Natural Experiment." *Journal of Political Economy* 117 (2): 257–80.
- Durlauf, Steven N., and Daniel S. Nagin. 2011. "Imprisonment and Crime: Can Both Be Reduced?" Criminology and Public Policy 10 (1): 13–54.
- Eck, John E., and Edward R. Maguire. 2000. "Have Changes in Policing Reduced Violent Crime? An Assessment of the Evidence." In *The Crime Drop in America*, edited by Alfred Blumstein and Joel Wallman, 207–65. Cambridge and New York: Cambridge University Press.
- Eck, John E., and David Weisburd. 1995. "Crime Places in Crime Theory." In *Crime and Place, Volume 4*, edited by John E. Eck and David Weisburd, 1–33. Monsey, NY: Criminal Justice Press; Washington, DC: Police Executive Research Forum.
- Ehrlich, Isaac. 1973. "Participation in Illegitimate Activities: A Theoretical and Empirical Investigation." *Journal of Political Economy* 81 (3): 521–65.
- Engel, Robin S., Nicholas Corsaro, and Marie Shubak Tillyer. 2010. "Evaluation of the Cincinnati Initiative to Reduce Violence (CIRV)." University of Cincinnati Policing Institute 354.
- Engel, Robin S., Marie Skubak Tillyer, and Nicholas Corsaro. 2013. "Reducing Gang Violence Using Focused Deterrence: Evaluating the Cincinnati Initiative to Reduce Violence (CIRV)." Justice Quarterly 30 (3): 403–39.
- Entorf, Horst, and Hannes Spengler. 2000. "Criminality, Social Cohesion and Economic Performance." Würzburg Economic Paper 22.
- Evans, William N., and Emily G. Owens. 2007. "COPS and Crime." *Journal of Public Economics* 91 (1–2): 181–201.
- Fagan, Jeffrey A., Amanda Geller, Garth Davies, and Valerie West. 2009. "Street Stops and Broken

Windows Revisited: The Demography and Logic of Proactive Policing in a Safe and Changing City." In *Race, Ethnicity, and Policing: New and Essential Readings*, edited by Stephen K. Rice and Michael D. White, 309–48. New York and London: New York University Press.

- Fernandez, Jose, Thomas Holman, and John V. Pepper. 2014. "The Impact of Living-Wage Ordinances on Urban Crime." *Industrial Relations* 53 (3): 478–500.
- Fougére, Denis, Francis Kramarz, and Julien Pouget. 2009. "Youth Unemployment and Crime in France." *Journal of the European Economic Association* 7 (5): 909–38.
- Freeman, Richard. 1983. "Crime and Unemployment." In Crime and Public Policy, edited by James Q. Wilson. San Francisco: ICS Press.
- Gorr, Wilpen L., and YongJei Lee. 2015. "Early Warning System for Temporary Crime Hot Spots." Journal of Quantitative Criminology 31 (1): 25–47.
- Gould, Eric D., Bruce A. Weinberg, and David B. Mustard. 2002. "Crime Rates and Local Labor Market Opportunities in the United States: 1979–1997." *Review of Economics and Statistics* 84 (1): 45–61.
- Greenwood, Peter, and Angela Hawken. 2002. "An Assessment of the Effects of California's Three Strikes Law." Unpublished.
- Griliches, Zvi, and Ĵerry A. Hausman. 1986. "Errors in Variables in Panel Data." *Journal of Econometrics* 31 (1): 93–118.
- Grogger, Jeffrey. 1991. "Certainty vs. Severity of Punishment." *Economic Inquiry* 29 (2): 297–309.
- Grogger, Jeffrey. 1995. "The Effect of Arrests on the Employment and Earnings of Young Men." *Quarterly Journal of Economics* 110 (1): 51–71.
- Grogger, Jeffrey. 1998. "Market Wages and Youth Crime." Journal of Labor Economics 16 (4): 756–91.
- Gronau, Reuben. 1980. "Home Production—A Forgotten Industry." *Review of Economics and Statistics* 62 (3): 408–16.
- Hansen, Benjamin, and Glen R. Waddell. 2014. "Walk Like a Man: Do Juvenile Offenders Respond to Being Tried as Adults?" Unpublished.
- Hansen, Kirstine, and Stephen Machin. 2002. "Spatial Crime Patterns and the Introduction of the UK Minimum Wage." Oxford Bulletin of Economics and Statistics 64 (Supplement): 677–97.
- Harcourt, Bernard E., and Jens Ludwig. 2006. "Broken Windows: New Evidence from New York City and a Five-City Social Experiment." University of Chicago Law Review 73: 271–320.
- Hawken, Angela, and Mark A. R. Kleiman. 2009. "Managing Drug Involved Probationers with Swift and Certain Sanctions: Evaluating Hawaii's HOPE." Washington, DC: National Criminal Justice Reference Service.
- Helland, Eric, and Alexander Tabarrok. 2007. "Does Three Strikes Deter? A Nonparametric Estimation." *Journal of Human Resources* 42 (2): 309–30.
- Hjalmarsson, Randi. 2009a. "Crime and Expected Punishment: Changes in Perceptions at the Age of Criminal Majority." American Law and Economics Review

11 (1): 209-48.

- Hjalmarsson, Randi. 2009b. "Juvenile Jails: A Path to the Straight and Narrow or to Hardened Criminality?" *Journal of Law and Economics* 52 (4): 779–809.
- Hjalmarsson, Randi. 2012. "Can Executions Have a Short-Term Deterrence Effect on Non-felony Homicides?" Criminology and Public Policy 11 (3): 565–71.
- Hope, Timothy J. 1994. "Problem-Oriented Policing and Drug-Market Locations: Three Case Studies." *Crime Prevention Studies* 2 (1): 5–32.
- Huang, Chien-Chieh, Derek Laing, and Ping Wang. 2004. "Crime and Poverty: A Search-Theoretic Approach." *International Economic Review* 45 (3): 909–38.
- Ihlanfeldt, Keith R. 2007. "Neighborhood Drug Crime and Young Males' Job Accessibility." *Review of Eco*nomics and Statistics 89 (1): 151–64.
- Imai, Susumu, and Kala Krishna. 2004. "Employment, Deterrence, and Crime in a Dynamic Model." *International Economic Review* 45 (3): 845–72.
- Johnson, Rucker, and Steven Raphael. 2012. "How Much Crime Reduction Does the Marginal Prisoner Buy?" *Journal of Law and Economics* 55 (2): 275–310.
- Katz, Lawrence, Steven D. Levitt, and Ellen Shustorovich. 2003. "Prison Conditions, Capital Punishment, and Deterrence." *American Law and Economics Review* 5 (2): 318–43.
- Kelling, George L., and William J. Bratton. 2015. "Why We Need Broken Windows Policing." City Journal Winter.
- Kelling, George L., and William H. Sousa, Jr. 2001. "Do Police Matter? An Analysis of the Impact of New York City's Policy Reforms." Manhattan Institute Center for Civic Innovation Civic Report 22.
- Kennedy, David M., Anthony A. Braga, Anne M. Piehl, and Elin J. Waring. 2001. "Reducing Gun Violence: The Boston Gun Project's Operation Ceasefire." US Department of Justice National Institute of Justice Research Report 188741.
- Kessler, Daniel, and Steven D. Levitt. 1999. "Using Sentence Enhancements to Distinguish between Deterrence and Incapacitation." *Journal of Law and Economics* 42 (Supplement 1): 343–64.
- Kilmer, Beau, Nancy Nicosia, Paul Heaton, and Greg Midgette. 2013. "Efficacy of Frequent Monitoring with Swift, Certain, and Modest Sanctions for Violations: Insights from South Dakota's 24/7 Sobriety Project." American Journal of Public Health 103 (1): 37–43.
- Kleck, Gary, and J. C. Barnes. 2014. "Do More Police Lead to More Crime Deterrence?" *Crime and Delinquency* 60 (5): 716–38.
- Kleck, Gary, Brion Sever, Spencer Li, and Marc Gertz. 2005. "The Missing Link in General Deterrence Research." Criminology 43 (3): 623–60.
- Kleiman, Mark A. R. 2009. When Brute Force Fails: How to Have Less Crime and Less Punishment. Princeton and Oxford: Princeton University Press.
- Klick, Jonathan, and Alexander Tabarrok. 2005. "Using Terror Alert Levels to Estimate the Effect of Police

on Crime." Journal of Law and Economics 48 (1): 267–79.

- Kovandzic, Tomislav V., Lynne M. Vieraitis, and Denise Paquette Boots. 2009. "Does the Death Penalty Save Lives? New Evidence from State Panel Data, 1977 to 2006." Criminology and Public Policy 8 (4): 803–43.
- Kubrin, Charis E., Steven F. Messner, Glenn Deane, Kelly McGeever, and Thomas D. Stucky. 2010. "Proactive Policing and Robbery Rates across U.S. Cities." Criminology 48 (1): 57–97.
- Land, Kenneth C., Raymond H. C. Teske, Jr., and Hui Zheng. 2009. "The Short-Term Effects of Executions on Homicides: Deterrence, Displacement, or Both?" *Criminology* 47 (4): 1009–43.
- Lattimore, Pamela K., Christy A. Visher, and Danielle M. Steffey. 2008. "Pre-release Characteristics and Service Receipt among Adult Male Participants in the SVORI Multi-site Evaluation." Washington, DC: Urban Institute.
- Lee, David S., and Justin McCrary. Forthcoming. "The Deterrence Effect of Prison: Dynamic Theory and Evidence." *Advances in Econometrics*.
- Lemieux, Thomas, Bernard Fortin, and Pierre Frechette. 1994. "The Effect of Taxes on Labor Supply in the Underground Economy." *American Economic Review* 84 (1): 231–54.
- Levitt, Steven D. 1996. "The Effect of Prison Population Size on Crime Rates: Evidence from Prison Overcrowding Litigation." *Quarterly Journal of Economics* 111 (2): 319–51.
- Levitt, Steven D. 1997. "Using Electoral Cycles in Police Hiring to Estimate the Effects of Police on Crime." American Economic Review 87 (3): 270–90.
- Levitt, Steven D. 1998. "Juvenile Crime and Punishment." *Journal of Political Economy* 106 (6): 1156–85.
- Levitt, Steven D. 2001. "Alternative Strategies for Identifying the Link Between Unemployment and Crime." *Journal of Quantitative Criminology* 17 (4): 377–90.
- Levitt, Steven D. 2002. "Using Electoral Cycles in Police Hiring to Estimate the Effects of Police on Crime: Reply." American Economic Review 92 (4): 1244–50.
- Levitt, Steven D. 2006. "The Case of the Critics Who Missed the Point: A Reply to Webster et al." Criminology and Public Policy 5 (3): 449–60.
- Levitt, Steven D., and Thomas J. Miles. 2006. "Economic Contributions to the Understanding of Crime." Annual Review of Law and Social Science 2: 147–64.
- Liedka, Raymond V., Anne Morrison Piehl, and Bert Useem. 2006. "The Crime-Control Effect of Incarceration: Does Scale Matter?" *Criminology and Public Policy* 5 (2): 245–76.
- Lin, Ming-Jen. 2008. "Does Unemployment Increase Crime? Evidence from U.S. Data 1974–2000." *Jour*nal of Human Resources 43 (2): 413–36.
- Lin, Ming-Jen. 2009. "More Police, Less Crime: Evidence from US State Data." *International Review of Law and Economics* 29 (2): 73–80.

- Lochner, Lance. 2007. "Individual Perceptions of the Criminal Justice System." American Economic Review 97 (1): 444–60.
- Lochner, Lance, and Enrico Moretti. 2004. "The Effect of Education on Crime: Evidence from Prison Inmates, Arrests, and Self-Reports." *American Economic Review* 94 (1): 155–89.
- Lofstrom, Magnus, and Steven Raphael. 2013. "Public Safety Realignment and Crime Rates in California." San Francisco: Public Policy Institute of California.
- Loftin, Colin, Milton Heumann, and David McDowall. 1983. "Mandatory Sentencing and Firearms Violence: Evaluating an Alternative to Gun Control." *Law and Society Review* 17 (2): 287–318.
- Loftin, Colin, and David McDowall. 1981. "One with a Gun Gets You Two': Mandatory Sentencing and Firearms Violence in Detroit." Annals of the American Academy of Political and Social Science 455: 150–67.
- Loftin, Colin, and David McDowall. 1984. "The Deterrent Effects of the Florida Felony Firearm Law." *Journal of Criminal Law and Criminology* 75 (1): 250–59.
- MacCoun, Robert, Rosalie Liccardo Pacula, Jamie Chriqui, Katherine Harris, and Peter Reuter. 2009.
 "Do Citizens Know Whether Their State Has Decriminalized Marijuana? Assessing the Perceptual Component of Deterrence Theory." *Review of Law* and Economics 5 (1): 347–71.
- MacDonald, John M. 2002. "The Effectiveness of Community Policing in Reducing Urban Violence." *Crime and Delinquency* 48 (4): 592–618.
- MacDonald, John M., Jonathan Klick, and Ben Grunwald. 2016. "The Effect of Private Police on Crime: Evidence from a Geographic Regression Discontinuity Design." *Journal of the Royal Statistical Society: Series A* 179 (3): 831–46.
- Machin, Stephen, and Olivier Marie. 2011. "Crime and Police Resources: The Street Crime Initiative." *Journal of the European Economic Association* 9 (4): 678–701.
- Machin, Stephen, and Costas Meghir. 2004. "Crime and Economic Incentives." *Journal of Human Resources* 39 (4): 958–79.
- Mallar, Charles D., and Craig V. D. Thornton. 1978. "Transitional Aid for Released Prisoners: Evidence from the Life Experiment." *Journal of Human Resources* 13 (2): 208–36.
- Maltz, Michael D., and Joseph Targonski. 2002. "A Note on the Use of County-Level UCR Data." *Journal of Quantitative Criminology* 18 (3): 297–318.
- Marvell, Thomas B., and Carlisle E. Moody, Jr. 1994. "Prison Population Growth and Crime Reduction." *Journal of Quantitative Criminology* 10 (2): 109–40.
- Marvell, Thomas B., and Carlisle E. Moody, Jr. 1996. "Specification Problems, Police Levels, and Crime Rates." *Criminology* 34 (4): 609–46.
- Mastrobuoni, Giovanni. 2013. "Police and Clearance Rates: Evidence from Recurrent Redeployments within a City." Unpublished.
- Matsueda, Ross L., Derek A. Kreager, and David Huizinga. 2006. "Deterring Delinquents: A Rational

Choice Model of Theft and Violence." American Sociological Review 71: 95–122.

- McCrary, Justin. 2002. "Using Electoral Cycles in Police Hiring to Estimate the Effect of Police on Crime: Comment." *American Economic Review* 92 (4): 1236–43.
- McCrary, Justin. 2010. "Dynamic Perspectives on Crime." In *Handbook on the Economics of Crime*, edited by Bruce L. Benson and Paul R. Zimmerman, 82–108. Cheltenham, UK and Northampton, MA: Edward Elgar.
- McDowall, David, Colin Loftin, and Brian Wiersema. 1992. "A Comparative Study of the Preventive Effects of Mandatory Sentencing Laws for Gun Crimes." *Journal of Criminal Law and Criminology* 83 (2): 378–94.
- McGarrell, Edmund F., Steven Chermak, Alexander Weiss, and Jeremy Wilson. 2001. "Reducing Firearms Violence through Directed Police Patrol." *Criminology and Public Policy* 1 (1): 119–48.
- McGarrell, Edmund F., Steven Chermak, Jeremy Wilson, and Nicholas Corsaro. 2006. "Reducing Homicide through a 'Lever-Pulling' Strategy." Justice Quarterly 23 (2): 214–31.
- McGarrell, Edmund F., Nicholas Corsaro, Natalie Kroovand Hipple, and Timothy S. Bynum. 2010. "Project Safe Neighborhoods and Violent Crime Trends in US Cities: Assessing Violent Crime Impact." Journal of Quantitative Criminology 26 (2): 165–90.
- McGarrell, Edmund F., et al. 2009. "Project Safe Neighborhoods—A National Program to Reduce Gun Crime: Final Project Report." Washington, DC: US Department of Justice National Institute of Justice.
- Mocan, H. Naci, and R. Kaj Gittings. 2003. "Getting Off Death Row: Commuted Sentences and the Deterrent Effect of Capital Punishment." *Journal of Law and Economics* 46 (2): 453–78.
- Mocan, H. Naci, and R. Kaj Gittings. 2010. "The Impact of Incentives on Human Behavior: Can We Make It Disappear? The Case of the Death Penalty." In *The Economics of Crime: Lessons For and From Latin America*, edited by Rafael Di Tella, Sebastian Edwards, and Ernesto Schargrodsky, 379–420. Chicago and London: University of Chicago Press.
- Mustard, David B. 2010. "How Do Labor Markets Affect Crime? New Evidence on an Old Puzzle." Institute for the Study of Labor Discussion Paper 4856.
- Nagin, Daniel S. 1998. "Criminal Deterrence Research at the Outset of the Twenty-First Century." Crime and Justice: A Review of Research 23: 1–42.
- Nagin, Daniel S. 2013. "Deterrence: A Review of the Evidence By a Criminologist for Economists." *Annual Review of Economics* 5: 83–105.
- Nagin, Daniel S., Francis T. Cullen, and Cheryl Lero Jonson. 2009. "Imprisonment and Re-offending." *Crime and Justice: A Review of Research* 38: 115–200.
- Nagin, Daniel S., and Raymond Paternoster. 1994. "Personal Capital and Social Control: The Deterrence Implications of a Theory of Individual Differences in

Criminal Offending." Criminology 32 (4): 581–606.

- Nagin, Daniel S., and Greg Pogarsky. 2001. "Integrating Celerity, Impulsivity, and Extralegal Sanction Threats into a Model of General Deterrence: Theory and Evidence." Criminology 39 (4): 865–92.
- Nagin, Daniel S., and Greg Pogarsky. 2003. "An Experimental Investigation of Deterrence: Cheating, Self-Serving Bias, and Impulsivity." *Criminology* 41 (1): 167–94.
- Nagin, Daniel S., and G. Matthew Snodgrass. 2013. "The Effect of Incarceration on Re-offending: Evidence from a Natural Experiment in Pennsylvania." *Journal of Quantitative Criminology* 29 (4): 601–42.
- Owens, Emily Greene. 2013. "COPS and Cuffs." In Lessons from the Economics of Crime: What Reduces Offending², edited by Philip J. Cook, Stephen Machin, Olivier Marie, and Giovanni Mastrobuoni, 17–44. Cambridge, MA and London: MIT Press.
- Papachristos, Andrew V., Tracey L. Meares, and Jeffrey Fagan. 2007. "Attention Felons: Evaluating Project Safe Neighborhoods in Chicago." *Journal of Empiri*cal Legal Studies 4 (2): 223–72.
- Papps, Kerry, and Rainer Winkelmann. 2000. "Unemployment and Crime: New Evidence for an Old Question." New Zealand Economic Papers 34 (1): 53–71.
- Paternoster, Raymond, and Alex Piquero. 1995. "Reconceptualizing Deterrence: An Empirical Test of Personal and Vicarious Experiences." *Journal of Research in Crime and Delinquency* 32 (3): 251–86.
- Petersilia, Joan, and Elizabeth Piper Deschenes. 1994. "Perceptions of Punishment: Inmates and Staff Rank the Severity of Prison versus Intermediate Sanctions." Prison Journal 74 (3): 306–28.
- Phillips, Julie, and Kenneth C. Land. 2012. "The Link between Unemployment and Crime Rate Fluctuations: An Analysis at the County, State, and National Levels." Social Science Research 41 (3): 681–94.
- Piehl, Anne Morrison. 1998. "Economic Conditions, Work, and Crime." In *The Handbook of Crime* and Punishment, edited by Michael Tonry, 302–22. Oxford and New York: Oxford University Press.
- Piquero, Alex R., and Greg Pogarsky. 2002. "Beyond Stafford and Warr's Reconceptualization of Deterrence: Personal and Vicarious Experiences, Impulsivity, and Offending Behavior." *Journal of Research in Crime and Delinquency* 39 (2): 153–86.
- Pogarsky, Greg, KiDeuk Kim, and Ray Paternoster. 2005. "Perceptual Change in the National Youth Survey: Lessons for Deterrence Theory and Offender Decision-Making." *Justice Quarterly* 22 (1): 1–29.
- Pogarsky, Greg, and Alex R. Piquero. 2003. "Can Punishment Encourage Offending? Investigating the 'Resetting' Effect." *Journal of Research in Crime and Delinquency* 40 (1): 95–120.
- Pogarsky, Greg, Alex R. Piquero, and Ray Paternoster. 2004. "Modeling Change in Perceptions about Sanction Threats: The Neglected Linkage in Deterrence Theory." *Journal of Quantitative Criminology* 20 (4): 343–69.
- Polinsky, A. Mitchell, and Steven Shavell. 1999. "On

the Disutility and Discounting of Imprisonment and the Theory of Deterrence." *Journal of Legal Studies* 28 (1): 1–16.

- Raphael, Steven. 2006. "The Deterrent Effects of California's Proposition 8: Weighing the Evidence." *Criminology and Public Policy* 5 (3): 471–78.
- Raphael, Steven, and Jens Ludwig. 2003. "Prison Sentence Enhancements: The Case of Project Exile." In *Evaluating Gun Policy: Effects on Crime and Violence*, edited by Jens Ludwig and Philip J. Cook, 251–76. Washington, DC: Brookings Institution Press.
- Raphael, Steven, and Rudolf Winter-Ebmer. 2001. "Identifying the Effect of Unemployment on Crime." *Journal of Law and Economics* 44 (1): 259–83.
- Roman, Caterina G., and Shannon E. Reid. 2012. "Assessing the Relationship between Alcohol Outlets and Domestic Violence: Routine Activities and the Neighborhood Environment." *Violence and Victims* 27 (5): 811–28.
- Rosenfeld, Richard, and Robert Fornango. 2007. "The Impact of Economic Conditions on Robbery and Property Crime: The Role of Consumer Sentiment." *Criminology* 45 (4): 735–69.
- Rosenfeld, Richard, Robert Fornango, and Eric Baumer. 2005. "Did *Ceasefire, Compstat*, and *Exile* Reduce Homicide?" *Criminology and Public Policy* 4 (3): 419–49.
- Rosenfeld, Richard, Robert Fornango, and Andres F. Rengifo. 2007. "The Impact of Order-Maintenance Policing on New York City Homicide and Robbery Rates: 1988–2001." Criminology 45 (2): 355–84.
- Rossi, Peter H., Richard A. Berk, and Kenneth J. Lenihan. 1980. Money, Work, and Crime: Experimental Evidence. New York and London: Academic Press.
- Sampson, Robert J., and Jacqueline Cohen. 1988. "Deterrent Effects of the Police on Crime: A Replication and Theoretical Extension." *Law and Society Review* 22 (1): 163–90.
- Schnepel, Kevin T. 2013. "Labor Market Opportunities and Crime: Evidence from Parolees." Unpublished.
- Shaw, Clifford R., and Henry D. McKay. 1942. Juvenile Delinquency and Urban Areas. Chicago: University of Chicago Press.
- Shepherd, Joanna M. 2002. "Fear of the First Strike: The Full Deterrent Effect of California's Two- and Three-Strikes Legislation." *Journal of Legal Studies* 31 (1): 159–201.
- Sherman, Lawrence W. 1990. "Police Crackdowns: Initial and Residual Deterrence." Crime and Justice 12: 1–48.
- Sherman, Lawrence W. 1995. "Hot Spots of Crime and Criminal Careers of Places." In *Crime and Place: Crime Prevention Studies*, Vol. 1, edited by John E. Eck and David Weisburd, 35–52. Monsey, NY: Willow Tree Press.
- Sherman, Lawrence W., et al. 1995. "Deterrent Effects of Police Raids on Crack Houses: A Randomized, Controlled Experiment." *Justice Quarterly* 12 (4): 755–81.
- Sherman, Lawrence W., Patrick R. Gartin, and Michael E. Buerger. 1989. "Hot Spots of Predatory

Crime: Routine Activities and the Criminology of Place." Criminology 27 (1): 27–56.

- Sherman, Lawrence W., and Dennis P. Rogan. 1995. "Effects of Gun Seizures on Gun Violence: 'Hot Spots' Patrol in Kansas City." *Justice Quarterly* 12 (4): 673–93.
- Sherman, Lawrence W., and David Weisburd. 1995. "General Deterrent Effects of Police Patrol in Crime 'Hot Spots': A Randomized, Controlled Trial." *Justice Quarterly* 12 (4): 625–48.
- Shi, Lan. 2009. "The Limit of Oversight in Policing : Evidence from the 2001 Cincinnati Riot." *Journal of Public Economics* 93 (1–2): 99–113.
- Skogan, Wesley, and Kathleen Frydl, eds. 2004. Fairness and Effectiveness in Policing: The Evidence. Washington, DC: National Academies Press.
- Smith, Adam. 1776. An Inquiry into the Nature and Causes of the Wealth of Nations. New York: Modern Library, 1937.
- Spelman, William. 2000. "What Recent Studies Do (and Don't) Tell Us about Imprisonment and Crime." *Crime and Justice: A Review of Research* 27: 419–94.
- Spelman, William, and Dale K. Brown. 1981. Calling the Police: Citizen Reporting of Serious Crime. Washington, DC: Police Executive Research Forum.
- Stolzenberg, Lisa, and Stewart J. D'Alessio. 1997. "Three Strikes and You're Out': The Impact of California's New Mandatory Sentencing Law on Serious Crime Rates." *Crime and Delinquency* 43 (4): 457–69.
- Stolzenberg, Lisa, and Stewart J. D'Alessio. 2004.
 "Capital Punishment, Execution Publicity and Murder in Houston, Texas." *Journal of Criminal Law and Criminology* 94 (2): 351–80.
 Thornberry, Terence P., and R. L. Christenson. 1984.
- Thornberry, Terence P., and R. L. Christenson. 1984. "Unemployment and Criminal Involvement: An Investigation of Reciprocal Causal Structures." *American Sociological Review* 49 (3): 398–411.
- Tonry, Michael. 2008. "Learning from the Limitations of Deterrence Research." *Crime and Justice* 37 (1): 279–311.
- Uggen, Christopher. 2000. "Work as a Turning Point in the Life Course of Criminals: A Duration Model of Age, Employment, and Recidivism." *American Sociological Review* 67: 529–46.
- Visher, Christy A., Laura Winterfield, and Mark B. Coggeshall. 2005. "Ex-offender Employment Programs and Recidivism: A Meta-analysis." *Journal of Experimental Criminology* 1 (3): 295–316.
- Waldo, Gordon P., and Theodore G. Chiricos. 1972. "Perceived Penal Sanction and Self-Reported Criminality: A Neglected Approach to Deterrence Research." Social Problems 19 (4): 522–40.
- Webster, Cheryl Marie, Anthony N. Doob, and Franklin E. Zimring. 2006. "Proposition 8 and Crime Rates in California: The Case of the Disappearing Deterrent." *Criminology and Public Policy* 5 (3): 417–48.
- Weisburd, David. 2005. "Hot Spots Policing Experiments and Criminal Justice Research: Lessons from the Field." ANNALS of the American Academy of Political and Social Science 599 (1): 220–45.

- Weisburd, David, Gerben J. N. Bruinsma, and Wim Bernasco. 2009. "Units of Analysis in Geographic Criminology: Historical Development, Critical Issues, and Open Questions." In Putting Crime in Its Place: Units of Analysis in Geographic Criminology, edited by David Weisburd, Wim Bernasco, and Gerben J. N. Bruinsma, 3–31. New York: Springer.
- Weisburd, David, Shawn Bushway, Cynthia Lum, and Sue-Ming Yang. 2004. "Trajectories of Crime at Places: A Longitudinal Study of Street Segments in the City of Seattle." Criminology 42 (2): 283–322.
- Weisburd, David, and Lorraine Green. 1995. "Policing Drug Hot Spots: The Jersey City Drug Market Analysis Experiment." Justice Quarterly 12 (4): 711–35.
- Weisburd, David, Lisa Maher, and Lawrence Sherman. 1992. "Contrasting Crime General and Crime Specific Theory: The Case of Hot Spots of Crime." In New Directions in Criminological Theory: Advances in Criminological Theory: Volume 4, edited by Freda Adler and William S. Laufer, 45–70. New Brunswick, NJ and London: Transaction Publishers.
- Weisburd, David, Stephen D. Mastrofski, Ann Marie McNally, Rosann Greenspan, and James J. Willis. 2003. "Reforming to Preserve: Compstat and Strategic Problem Solving in American Policing." Criminology and Public Policy 2 (3): 421–56.
- Weisburd, David, Nancy A. Morris, and Elizabeth R. Groff. 2009. "Hot Spots of Juvenile Crime: A Longitudinal Study of Arrest Incidents at Street Segments in Seattle, Washington." Journal of Quantitative Criminology 25: 443–67.
- Weisburd, David, Cody W. Telep, Joshua C. Hinkle, and John E. Eck. 2010. "Is Problem-Oriented Policing Effective in Reducing Crime and Disorder? Findings from a Campbell Systematic Review." Criminology and Public Policy 9 (1): 139–72.
- Weisburd, David, Laura A. Wyckoff, Justin Ready, John E. Eck, Joshua C. Hinkle, and Frank Gajewski. 2006. "Does Crime Just Move around the Corner? A Controlled Study of Spatial Displacement and Diffusion of Crime Control Benefits." *Criminology* 44 (3): 549–92.
- Williams, Jenny, and Robin C. Sickles. 2002. "An

Analysis of the Crime as Work Model: Evidence from the 1958 Philadelphia Birth Cohort Study." *Journal* of Human Resources 37 (3): 479–509.

- Williams, Kirk R., and Richard Hawkins. 1986. "Perceptual Research on General Deterrence: A Critical Review." *Law and Society Review* 20 (4): 545–72.
- Wilson, James Q., and Barbara Boland. 1978. "The Effect of the Police on Crime." *Law and Society Review* 12 (3): 367–90.
- Wilson, James Q., and George L. Kelling. 1982. "Broken Windows: The Police and Neighborhood Safety." *Atlantic Monthly* 249 (3): 29–38.
- Witte, Ann Dryden. 1980. "Estimating the Economic Model of Crime with Individual Data." *Quarterly Journal of Economics* 94 (1): 57–84.
- Witte, Ann Dryden, and Helen Tauchen. 1993. "Work and Crime: An Exploration Using Panel Data." In *The Economic Dimensions of Crime*, edited by Nigel G. Fielding, Alan Clarke, and Robert Witt, 177–92. London: MacMillan.
- Worrall, John L., and Tomislav V. Kovandzic. 2007. "COPS Grants and Crime Revisited." *Criminology* 45 (1): 159–90.
- Zimmerman, Paul R. 2004. "State Executions, Deterrence, and the Incidence of Murder." *Journal of Applied Economics* 7 (1): 163–93.
- Zimmerman, Paul R. 2006. "Estimates of the Deterrent Effect of Alternative Execution Methods in the United States: 1978–2000." American Journal of Economics and Sociology 65 (4): 909–41.
- Zimmerman, Paul R. 2009. "Statistical Variability and the Deterrent Effect of the Death Penalty." American Law and Economics Review 11 (2): 370–98.
- Zimring, Franklin E., Jeffrey Fagan, and David T. Johnson. 2010. "Executions, Deterrence, and Homicide: A Tale of Two Cities." *Journal of Empirical Legal Studies* 7 (1): 1–29.
- Zimring, Franklin E., and Gordon Hawkins. 1973. Deterrence: The Legal Threat in Crime Control. Chicago and London: University of Chicago Press.
- Zimring, Franklin E., Gordon Hawkins, and Sam Kamin. 2001. Punishment and Democracy: Three Strikes and You're Out in California. Oxford and New York: Oxford University Press.

This article has been cited by:

- 1. Qing Zhu, Fan Zhang, Shan Liu, Lin Wang, Shouyang Wang. 2022. Static or dynamic? Characterize and forecast the evolution of urban crime distribution. *Expert Systems with Applications* **190**, 116115. [Crossref]
- 2. Matthias Kasper, James Alm. 2022. Audits, audit effectiveness, and post-audit tax compliance. *Journal of Economic Behavior & Organization* 195, 87-102. [Crossref]
- 3. Esther Nir, Siyu Liu. 2022. The Challenge of Imposing Just Sentences Under Mandatory Minimum Statutes: A Qualitative Study of Judicial Perceptions. *Criminal Justice Policy Review* 33:2, 177-205. [Crossref]
- 4. Sebastian Blesse, André Diegmann. 2022. The place-based effects of police stations on crime: Evidence from station closures. *Journal of Public Economics* 207, 104605. [Crossref]
- 5. Devah Pager, Rebecca Goldstein, Helen Ho, Bruce Western. 2022. Criminalizing Poverty: The Consequences of Court Fees in a Randomized Experiment. *American Sociological Review* 15, 000312242210757. [Crossref]
- 6. John Braithwaite. Macrocriminology and Freedom . [Crossref]
- 7. Thomas Epper, Ernst Fehr, Kristoffer Balle Hvidberg, Claus Thustrup Kreiner, Søren Leth-Petersen, Gregers Nytoft Rasmussen. 2022. Preferences predict who commits crime among young men. *Proceedings of the National Academy of Sciences* **119**:6, e2112645119. [Crossref]
- Ashlee Curtis, Clare Farmer, Travis Harries, Richelle Mayshak, Kerri Coomber, Belinda Guadagno, Peter Miller. 2022. Do patron bans act as a deterrent to future anti-social offending? An analysis of banning and offending data from Victoria, Australia. *Policing and Society* 32:2, 234-247. [Crossref]
- Franziska M. Yasrebi-De Kom, Anja J. E. Dirkzwager, Peter H. Van Der Laan, Paul Nieuwbeerta. 2022. The Effect of Sanction Severity and Its Interaction With Procedural Justice. *Criminal Justice and Behavior* 49:2, 200-219. [Crossref]
- 10. Michael D. Noel. 2022. Competitive survival in a devastated industry: Evidence from hotels during COVID-19. Journal of Economics & Management Strategy 31:1, 3-24. [Crossref]
- Penélope Hernández, Zvika Neeman. 2022. How Bayesian Persuasion Can Help Reduce Illegal Parking and Other Socially Undesirable Behavior. *American Economic Journal: Microeconomics* 14:1, 186-215. [Abstract] [View PDF article] [PDF with links]
- 12. M. Antonella Mancino. 2022. A SEARCH MODEL OF EARLY EMPLOYMENT CAREERS AND YOUTH CRIME. *International Economic Review* **63**:1, 329-390. [Crossref]
- 13. Claude Fluet, Murat C Mungan. 2022. Reputational versus Beckerian Sanctions. American Law and Economics Review 115. . [Crossref]
- 14. Matus Halas. 2022. NATO's sub-conventional deterrence: The case of Russian violations of the Estonian airspace. *Contemporary Security Policy* **34**, 1-32. [Crossref]
- 15. Benjamin A. Howell, Valerie A. Earnshaw, Marisol Garcia, Andrew Taylor, Karin Martin, Aaron D. Fox. 2022. The Stigma of Criminal Legal Involvement and Health: a Conceptual Framework. *Journal of Urban Health* **389**. [Crossref]
- 16. C Matthew Leister, Yves Zenou, Junjie Zhou. 2022. Social Connectedness and Local Contagion. *The Review of Economic Studies* **89**:1, 372-410. [Crossref]

- 17. Anna Bindler, Randi Hjalmarsson. 2021. The Impact of the First Professional Police Forces on Crime. *Journal of the European Economic Association* **19**:6, 3063–3103. [Crossref]
- A Mitchell Polinsky, Steven Shavell. 2021. Deterrence and the Adjustment of Sentences During Imprisonment. American Law and Economics Review 23:2, 481-519. [Crossref]
- 19. Murat C Mungan. 2021. Rewards versus Imprisonment. American Law and Economics Review 23:2, 432-480. [Crossref]
- 20. Süleyman Emre ÖZCAN, Feride Özge ŞİRİN. 2021. Ekonomik Suçlar, Gelir ve İşsizliğin Mekânsal İlişkisi: 2014 Türkiye İncelemesi. *Celal Bayar Üniversitesi Sosyal Bilimler Dergisi* 79-100. [Crossref]
- 21. Anna Bindler, Nadine Ketel. 2021. Scaring or Scarring? Labour Market Effects of Criminal Victimisation. *Journal of Labor Economics*. [Crossref]
- 22. Roee Sarel. 2021. Crime and punishment in times of pandemics. *European Journal of Law and Economics* 194. . [Crossref]
- 23. Franklin Nakpodia, Emmanuel Adegbite, Folajimi Ashiru. 2021. Corporate governance regulation: a practice theory perspective. *Accounting Forum* **6**, 1-26. [Crossref]
- 24. Pavel Teplyashin, Vyacheslav Molokov. 2021. Agglomerative Approach to Assessing the Degree of Criminalization of the Subjects of the Russian Federation: the Methodology of Statistical Analysis and Criminological Description. *Russian Journal of Criminology* 15:5, 509-521. [Crossref]
- 25. Tamara Stotz, Angela Bearth, Signe Maria Ghelfi, Michael Siegrist. 2021. Keep the status quo: randomization-based security checks might reduce crime deterrence at airports. *Journal of Risk Research* 24:12, 1589-1604. [Crossref]
- 26. Joanna J. J. Wang, Thomas Fung, Donald Weatherburn. 2021. The impact of the COVID-19, social distancing, and movement restrictions on crime in NSW, Australia. *Crime Science* 10:1. [Crossref]
- 27. Henk Harmsen, Judith Syombua Mbau, John Nzioka Muthama, Virginia Wangechi Wang'ondu. 2021. Comparing law enforcement monitoring data and research data suggests an underestimation of bushmeat poaching through snaring in a Kenyan World Heritage Site. *African Journal of Ecology* 59:4, 912-923. [Crossref]
- Matthias Van Hall, Laura Cleofa-van Der Zwet. 2021. Released from foreign detention: Examining reoffending rates among returning Dutch detainees in the Netherlands. *European Journal of Probation* 13:3, 242-261. [Crossref]
- 29. Quin Patterson, Michael D. White. 2021. Is There a Civilizing Effect on Citizens? Testing the Pre-Conditions for Body Worn Camera-Induced Behavior Change. *Police Quarterly* 24:4, 411-437. [Crossref]
- 30. Ricardo Massa, Gustavo Fondevila. 2021. Criminal displacement in Mexico city's metropolitan area: The case of kidnapping. *International Journal of Law, Crime and Justice* **67**, 100479. [Crossref]
- 31. Michelle Anagnostou. 2021. Synthesizing knowledge on crime convergence and the illegal wildlife trade. *Environmental Challenges* 5, 100222. [Crossref]
- 32. Damon M. Petrich, Travis C. Pratt, Cheryl Lero Jonson, Francis T. Cullen. 2021. Custodial Sanctions and Reoffending: A Meta-Analytic Review. *Crime and Justice* **50**:1, 353-424. [Crossref]
- 33. Matthew Renner, Bradley Bartos. 2021. Decarceration, Sanction Severity and Crime: Causal Analysis of Proposition 47 and Property Crime in Los Angeles. *Justice Evaluation Journal* 42, 1-27. [Crossref]

- 34. Frank Darkwa Baffour. 2021. Recidivism: Exploring Why Inmates Re-Offend in a Prison Facility in Ghana. *Victims & Offenders* 16:8, 1161-1181. [Crossref]
- 35. Philipp M. Dau, Christophe Vandeviver, Maite Dewinter, Frank Witlox, Tom Vander Beken. 2021. Policing Directions: a Systematic Review on the Effectiveness of Police Presence. *European Journal* on Criminal Policy and Research 15. [Crossref]
- 36. Kristina Block. 2021. Professional Sports and Crime: Do Professional Hockey Games Increase City-Level Crime Rates?. Crime & Delinquency 67:12, 2069-2087. [Crossref]
- 37. Greg Midgette, Thomas A. Loughran, Sarah Tahamont. 2021. The Impact of Ambiguity-induced Error in Offender Decision-making: Evidence from the Field. *Journal of Research in Crime and Delinquency* 58:6, 635-665. [Crossref]
- Kinneret Teodorescu, Ori Plonsky, Shahar Ayal, Rachel Barkan. 2021. Frequency of enforcement is more important than the severity of punishment in reducing violation behaviors. *Proceedings of the National Academy of Sciences* 118:42, e2108507118. [Crossref]
- 39. Federico Masera. 2021. Violent Crime and the Overmilitarization of US Policing. *The Journal of Law, Economics, and Organization* 37:3, 479-511. [Crossref]
- 40. Anne Sofie Tegner Anker, Jennifer L. Doleac, Rasmus Landersø. 2021. The Effects of DNA Databases on the Deterrence and Detection of Offenders. *American Economic Journal: Applied Economics* 13:4, 194-225. [Abstract] [View PDF article] [PDF with links]
- Yolanda Rebollo-Sanz, Jesús Rodríguez-López, Núria Rodríguez-Planas. 2021. Penalty-point system, deterrence and road safety: A quasi-experimental analysis. *Journal of Economic Behavior &* Organization 190, 408-433. [Crossref]
- 42. Matheus Albergaria. 2021. Every Book You Take: Evaluating Compliance Behavior in an Information Commons. *Estudos Econômicos (São Paulo)* 51:4, 643-675. [Crossref]
- 43. Kristian Skrede Gleditsch, Mauricio Rivera, Bárbara Zárate-Tenorio. 2021. Can Education Reduce Violent Crime? Evidence from Mexico before and after the Drug War Onset. *The Journal of Development Studies* 582, 1-18. [Crossref]
- 44. Libor Dušek, Christian Traxler. 2021. Learning from Law Enforcement. *Journal of the European Economic Association* 49. . [Crossref]
- 45. Verónica Michel. 2021. Institutional Design, Prosecutorial Independence, and Accountability: Lessons from the International Commission against Impunity in Guatemala (CICIG). *Laws* 10:3, 58. [Crossref]
- 46. Greg Midgette, Beau Kilmer, Nancy Nicosia, Paul Heaton. 2021. A Natural Experiment to Test the Effect of Sanction Certainty and Celerity on Substance-Impaired Driving: North Dakota's 24/7 Sobriety Program. *Journal of Quantitative Criminology* 37:3, 647-670. [Crossref]
- 47. Michael Alvard, Katherine Daiy. 2021. A Naturalistic Study of Norm Conformity, Punishment, and the Veneration of the Dead at Texas A&M University, USA. *Human Nature* **32**:3, 652-675. [Crossref]
- 48. Danielle E. Warren, Maurice E. Schweitzer. 2021. When weak sanctioning systems work: Evidence from auto insurance industry fraud investigations. *Organizational Behavior and Human Decision Processes* 166, 68-83. [Crossref]
- 49. Marco Francesconi, Jonathan James. 2021. None for the Road? Stricter Drink Driving Laws and Road Accidents. *Journal of Health Economics* **79**, 102487. [Crossref]

- Saron Chalfin, Michael LaForest, Jacob Kaplan. 2021. Can Precision Policing Reduce Gun Violence? Evidence from "Gang Takedowns" in New York City. *Journal of Policy Analysis and Management* 40:4, 1047-1082. [Crossref]
- 51. Christopher Blattman, Donald P Green, Daniel Ortega, Santiago Tobón. 2021. Place-Based Interventions at Scale: The Direct and Spillover Effects of Policing and City Services on Crime. *Journal* of the European Economic Association 19:4, 2022-2051. [Crossref]
- 52. Wenhui Zhu, Yuhang Zheng, Kunhui Ye, Qian Zhang, Minjie Zhang. 2021. Deterrence of Punitive Measures on Collusive Bidding in the Construction Sector. *Complexity* 2021, 1-12. [Crossref]
- Javier D. Donna, José-Antonio Espín-Sánchez. 2021. Water theft as social insurance: south-eastern Spain, 1851–1948 †. The Economic History Review 74:3, 721-753. [Crossref]
- 54. Samuel Bazzi, Gordon Hanson, Sarah John, Bryan Roberts, John Whitley. 2021. Deterring Illegal Entry: Migrant Sanctions and Recidivism in Border Apprehensions. *American Economic Journal: Economic Policy* 13:3, 1-27. [Abstract] [View PDF article] [PDF with links]
- 55. Fabio Galeotti, Valeria Maggian, Marie Claire Villeval. 2021. Fraud Deterrence Institutions Reduce Intrinsic Honesty. *The Economic Journal* 131:638, 2508-2528. [Crossref]
- 56. Lucia Dalla Pellegrina, Margherita Saraceno. 2021. Does the open-cell regime foster inmates' legal capability? Evidence from two Italian prisons. *European Journal of Law and Economics* 52:1, 89-135. [Crossref]
- 57. Sofie Pilemalm. Digitalized Co-production: Using Volunteers as First Responders 25-32. [Crossref]
- 58. Manuel Ramos Maqueda, Daniel L. Chen. The Role of Justice in Development: The Data Revolution 3, . [Crossref]
- 59. Angelo Castaldo, Anna Rita Germani, Antonio Pergolizzi. 2021. Does Education Affect Environmental Crime? A Dynamic Panel Data Approach at Provincial Level in Italy. *International Criminal Justice Review* 3, 105756772110208. [Crossref]
- 60. Don Weatherburn, Steve Yeong. 2021. Does correctional supervision of amphetamine users reduce the risk of re-offending?. *Addiction* 116:6, 1472-1481. [Crossref]
- 61. Eric Reinhart, Daniel L. Chen. 2021. Carceral-community epidemiology, structural racism, and COVID-19 disparities. *Proceedings of the National Academy of Sciences* 118:21, e2026577118. [Crossref]
- 62. Sarit Weisburd. 2021. Police Presence, Rapid Response Rates, and Crime Prevention. *The Review of Economics and Statistics* 103:2, 280-293. [Crossref]
- Felipe Goncalves, Steven Mello. 2021. A Few Bad Apples? Racial Bias in Policing. American Economic Review 111:5, 1406-1441. [Abstract] [View PDF article] [PDF with links]
- 64. Kocherlakota Satya Pritam, Sugandha, Trilok Mathur, Shivi Agarwal. 2021. Underlying dynamics of crime transmission with memory. *Chaos, Solitons & Fractals* 146, 110838. [Crossref]
- Emiliano Tealde. 2021. The unequal impact of natural light on crime. *Journal of Population Economics* 94. [Crossref]
- Jason L. Payne, Cameron T. Langfield. 2021. How risky are heroin markets? A multi-site study of self-reported risk perceptions among police detainees in Australia. *International Journal of Drug Policy* 90, 103062. [Crossref]

- 67. Johannes Buckenmaier, Eugen Dimant, Ann-Christin Posten, Ulrich Schmidt. 2021. Efficient Institutions and Effective Deterrence: On Timing and Uncertainty of Formal Sanctions. *Journal of Risk and Uncertainty* 62:2, 177-201. [Crossref]
- 68. Xi Chen, Hua Zhong. 2021. Development and Crime Drop: A Time-Series Analysis of Crime Rates in Hong Kong in the Last Three Decades. *International Journal of Offender Therapy and Comparative Criminology* 65:4, 409-433. [Crossref]
- 69. Tom Tyler, Matt Katsaros, Tracey Meares, Sudhir Venkatesh. 2021. Social media governance: can social media companies motivate voluntary rule following behavior among their users?. *Journal of Experimental Criminology* 17:1, 109-127. [Crossref]
- 70. Mukesh Eswaran. 2021. On Reducing the Sexual Assault of Women: What Can Economists Contribute to the Debate?. *Canadian Public Policy* 47:1, 117-129. [Crossref]
- David Mitre-Becerril, Aaron Chalfin. 2021. Testing public policy at the frontier: The effect of the \$15 minimum wage on public safety in Seattle. *Criminology & Public Policy* 84. . [Crossref]
- 72. Rosario Crinó, Giovanni Immordino, Salvatore Piccolo. 2021. Criminal mobility, fugitives, and extradition rules. *Journal of Public Economic Theory* 23:1, 69-104. [Crossref]
- David S. Abrams. 2021. COVID and crime: An early empirical look. *Journal of Public Economics* 194, 104344. [Crossref]
- 74. Aaron Chalfin, Benjamin Hansen, Jason Lerner, Lucie Parker. 2021. Reducing Crime Through Environmental Design: Evidence from a Randomized Experiment of Street Lighting in New York City. *Journal of Quantitative Criminology* 81. . [Crossref]
- 75. Alessandra Canepa, Federico Drogo. 2021. Wildfire crime, apprehension and social vulnerability in Italy. *Forest Policy and Economics* **122**, 102330. [Crossref]
- 76. Dhaval Dave, Monica Deza, Brady Horn. 2021. Prescription drug monitoring programs, opioid abuse, and crime. *Southern Economic Journal* 87:3, 808-848. [Crossref]
- 77. Shawn Kantor, Carl T. Kitchens, Steven Pawlowski. 2021. CIVIL ASSET FORFEITURE, CRIME, AND POLICE INCENTIVES: EVIDENCE FROM THE COMPREHENSIVE CRIME CONTROL ACT OF 1984. *Economic Inquiry* 59:1, 217-242. [Crossref]
- 78. Qi Ge, Ignacio Sarmiento Barbieri, Rodrigo Schneider. 2021. SPORTING EVENTS, EMOTIONAL CUES, AND CRIME: SPATIAL AND TEMPORAL EVIDENCE FROM BRAZILIAN SOCCER GAMES. *Economic Inquiry* 59:1, 375-395. [Crossref]
- 79. Georges Bresson, Guy Lacroix, Mohammad Arshad Rahman. 2021. Bayesian panel quantile regression for binary outcomes with correlated random effects: an application on crime recidivism in Canada. *Empirical Economics* **60**:1, 227-259. [Crossref]
- Sandra V. Rozo, Therese Anders, Steven Raphael. 2021. Deportation, crime, and victimization. *Journal of Population Economics* 34:1, 141-166. [Crossref]
- 81. Adolfo Figueroa. On Economic Behavior of Workers 89-110. [Crossref]
- 82. Theo Nyreröd, Giancarlo Spagnolo. 2021. A fresh look at whistleblower rewards. *Journal of Governance and Regulation* 10:4, special issue, 248-260. [Crossref]
- 83. Michelle Anagnostou, WilliamD Moreto, CharlieJ Gardner, Brent Doberstein. 2021. Poverty, Pandemics, and Wildlife Crime. *Conservation and Society* 19:4, 294. [Crossref]

- 84. Francesco Giumelli. Targeted Sanctions and Deterrence in the Twenty-first Century 349-363. [Crossref]
- 85. Marco Fabbri, Jonathan Klick. 2020. The Ineffectiveness of 'Observe and Report' Patrols on Crime. International Review of Law and Economics 105972. [Crossref]
- Helmut Hirtenlehner. 2020. Differenzielle Abschreckbarkeit als Evidenzgrundlage negativer Generalprävention – Eine Bestandsaufnahme der kriminologischen Wissensbasis. *Monatsschrift für* Kriminologie und Strafrechtsreform 103:3, 221-234. [Crossref]
- Rafael D.E.H. Castro, Martha Pena-Sarmiento, H. Norza C. Ervyn, Camilo A. Sanchez, Erick Gillen, Yeizon A. Duarte, Luis O. Jimenez A. Assessing of frequency dynamics of EEG signals in a visualization experiment related to crime deterrence 194-199. [Crossref]
- 88. Maurice J. G. Bun, Richard Kelaher, Vasilis Sarafidis, Don Weatherburn. 2020. Crime, deterrence and punishment revisited. *Empirical Economics* **59**:5, 2303-2333. [Crossref]
- 89. Lauren Wilson, Rachel Boratto. 2020. Conservation, wildlife crime, and tough-on-crime policies: Lessons from the criminological literature. *Biological Conservation* 251, 108810. [Crossref]
- 90. Robert Gonzalez, Sarah Komisarow. 2020. Community monitoring and crime: Evidence from Chicago's Safe Passage Program. *Journal of Public Economics* **191**, 104250. [Crossref]
- Xiaozhao Y. Yang, Brian Kelly, Tingzhong Yang. 2020. Peer Association and Routine Activities in Sex Worker Patronage among Male Migrant Workers. *Deviant Behavior* 4, 1-18. [Crossref]
- 92. Louis-Pierre Lepage. 2020. Are criminals strategic? Offender responses to drug sentencing cutoffs. Labour Economics 66, 101906. [Crossref]
- 93. Yuya Kudo. 2020. Maintaining law and order: Welfare implications from village vigilante groups in northern Tanzania. *Journal of Economic Behavior & Organization* 178, 607-628. [Crossref]
- 94. Alex Raskolnikov. 2020. Criminal Deterrence: A Review of the Missing Literature. Supreme Court Economic Review 28, 1-59. [Crossref]
- 95. Fernanda Martínez Flores. 2020. The Effects of Enhanced Enforcement at Mexico's Southern Border: Evidence From Central American Deportees. *Demography* **57**:5, 1597-1623. [Crossref]
- 96. Colleen M. Berryessa, Joel M. Caplan. 2020. Cognitive and Affective Processing of Risk Information: A Survey Experiment on Risk-Based Decision-Making Related to Crime and Public Safety. *Frontiers in Psychology* 11. [Crossref]
- 97. ELEFTHERIOS GOULAS, SOCRATES KARIDIS. 2020. CRIME AND FISCAL POLICY IN EUROPE: THE EFFECT OF SHADOW ECONOMY. *Global Economy Journal* 20:03, 2050017. [Crossref]
- 98. Stéphane Mechoulan. 2020. Civil unrest, emergency powers, and spillover effects: A mixed methods analysis of the 2005 French riots. *Journal of Economic Behavior & Organization* 177, 305-326. [Crossref]
- 99. Berno Buechel, Eberhard Feess, Gerd Muehlheusser. 2020. Optimal law enforcement with sophisticated and naïve offenders. *Journal of Economic Behavior & Organization* 177, 836-857. [Crossref]
- 100. Dietrich Earnhart. 2020. Stated choices of environmental managers: The role of punishment. *International Review of Law and Economics* 63, 105908. [Crossref]

- 101. Justin Gallagher, Paul J. Fisher. 2020. Criminal Deterrence when There Are Offsetting Risks: Traffic Cameras, Vehicular Accidents, and Public Safety. *American Economic Journal: Economic Policy* 12:3, 202-237. [Abstract] [View PDF article] [PDF with links]
- 102. Rocco d'Este. 2020. The Effects of Stolen-Goods Markets on Crime: Pawnshops, Property Theft, and the Gold Rush of the 2000s. *The Journal of Law and Economics* 63:3, 449-472. [Crossref]
- 103. Clare Farmer, Robyn Clifford. 2020. Spatial Exclusion, Due Process, and the Civilianization of Punishment in Australia's Night-time Economy: A Mapping Review of Patron Banning Policy, Practice, and Oversight. *International Criminal Justice Review* 3, 105756772093915. [Crossref]
- 104. Sarah B. Hunter, Mahlet Tebeka, Beau Kilmer. 2020. Implementing Frequent Substance Use Testing with Swift-Certain-Fair Sanctions: Stakeholder Insights from a Pilot Program with Volunteer Probationers. *Journal of Drug Policy Analysis*, ahead of print. [Crossref]
- 105. Néstor Gandelman, Ignacio Munyo. 2020. Juvenile incarceration and crime after release: short-term evidence from a Harsher law. *Journal of Development Effectiveness* 12:3, 240-254. [Crossref]
- 106. Olivia K. Ha, Martin A. Andresen, Garth Davies. 2020. The Temporal (In)Stability of the Unemployment and Crime Relationship. *International Journal of Offender Therapy and Comparative Criminology* 64:8, 840-859. [Crossref]
- 107. Andrew Leigh. 2020. The Second Convict Age: Explaining the Return of Mass Imprisonment in Australia. *Economic Record* 96:313, 187-208. [Crossref]
- 108. Beau Kilmer, Greg Midgette. 2020. Criminal Deterrence: Evidence from an Individual-Level Analysis of 24/7 Sobriety. *Journal of Policy Analysis and Management* **39**:3, 801-834. [Crossref]
- 109. Amedeo Argentiero, Bruno Chiarini, Elisabetta Marzano. 2020. Does Tax Evasion Affect Economic Crime?*. *Fiscal Studies* 41:2, 441-482. [Crossref]
- 110. Sebastian Galiani, Laura Jaitman, Federico Weinschelbaum. 2020. Crime and durable goods. Journal of Economic Behavior & Organization 173, 146-163. [Crossref]
- 111. Manudeep Bhuller, Gordon B. Dahl, Katrine V. Løken, Magne Mogstad. 2020. Incarceration, Recidivism, and Employment. *Journal of Political Economy* **128**:4, 1269-1324. [Crossref]
- 112. Francesco Drago, Friederike Mengel, Christian Traxler. 2020. Compliance Behavior in Networks: Evidence from a Field Experiment. *American Economic Journal: Applied Economics* 12:2, 96-133. [Abstract] [View PDF article] [PDF with links]
- 113. Tom Kirchmaier, Stephen Machin, Matteo Sandi, Robert Witt. 2020. Prices, Policing and Policy: The Dynamics of Crime Booms and Busts. *Journal of the European Economic Association* 18:2, 1040-1077. [Crossref]
- 114. Eugen Dimant, Gerben A. van Kleef, Shaul Shalvi. 2020. Requiem for a Nudge: Framing effects in nudging honesty. *Journal of Economic Behavior & Organization* 172, 247-266. [Crossref]
- 115. Vedran Recher. 2020. Unemployment and property crime: evidence from Croatia. *Crime, Law and Social Change* **73**:3, 357-376. [Crossref]
- 116. Laura Nahuelhual, Gonzalo Saavedra, María Amalia Mellado, Ximena Vergara Vergara, Tomás Vallejos. 2020. A social-ecological trap perspective to explain the emergence and persistence of illegal fishing in small-scale fisheries. *Maritime Studies* 19:1, 105-117. [Crossref]
- 117. Xinlin Peng, Mingyu Gong, Jie Bing. 2020. Sweeping and eliminating evils: a diachronic study of mafia in China. *International Journal of Legal Discourse* 4:2, 197-216. [Crossref]

- 118. Mauricio Leiva, Felipe Vasquez-Lavín, Roberto D. Ponce Oliva. 2020. Do immigrants increase crime? Spatial analysis in a middle-income country. *World Development* **126**, 104728. [Crossref]
- 119. Shaina Herman, Greg Pogarsky. 2020. Morality, Deterrability, and Offender Decision Making. *Justice Quarterly* **4**, 1-25. [Crossref]
- 120. Anna Rita Germani, Alan P. Ker, Angelo Castaldo. 2020. On the existence and shape of an environmental crime Kuznets Curve: A case study of Italian provinces. *Ecological Indicators* 108, 105685. [Crossref]
- 121. Juan Esteban Carranza Romero, Carlos Giovanni González Espitia, Gipsy Eliana Bocanegra Ochoa. 2020. Efecto de la actividad económica sobre la violencia homicida: Nueva evidencia a partir de datos de panel en Colombia. *Cuadernos de Economía* 39:79, 355-388. [Crossref]
- 122. Alexander Lundberg, Murat C. Mungan. 2020. The Effect of Evidentiary Rules on Conviction Rates. SSRN Electronic Journal . [Crossref]
- 123. Ramphul Ohlan. 2020. Are regional crime rates in India natural?. *Crime, Law and Social Change* 73:1, 93-110. [Crossref]
- 124. Anna Bindler, Nadine Ketel, Randi Hjalmarsson. Costs of Victimization 1-31. [Crossref]
- 125. Anna Bindler, Nadine Ketel, Randi Hjalmarsson. Costs of Victimization 1-31. [Crossref]
- 126. James A Foulds, David Nutt. 2020. Principled sentencing for drug supply offences: Revised methamphetamine sentencing guidelines in New Zealand. Drug Science, Policy and Law 6, 205032452094234. [Crossref]
- 127. Ruben Poblete-Cazenave. 2020. The Great Lockdown and Criminal Activity Evidence from Bihar, India. SSRN Electronic Journal. [Crossref]
- 128. Michael D. Noel. 2020. COVID Economics Competitive Responses in a Devastated Industry: Evidence from Hotels during COVID-19. SSRN Electronic Journal . [Crossref]
- 129. Robert M Gonzalez, Ranae Jabri, Sarah Komisarow. 2020. The Goldilocks Effect: Crime and Punishment within Chicago's Drug-Free School Zones. *SSRN Electronic Journal*. [Crossref]
- 130. Rustam Romaniuc, Eugen Dimant. 2020. Understanding Cross-Cultural Differences in Peer Reporting Practices: Evidence from Tax Evasion Games in Moldova and France. SSRN Electronic Journal. [Crossref]
- 131. Aditya Kuvalekar, Joao Ramos, Johannes Schneider. 2020. Intuition in the Shadow of the Court. SSRN Electronic Journal. [Crossref]
- Roee Sarel. 2020. Crime and Punishment in Times of Pandemics. SSRN Electronic Journal 113. . [Crossref]
- Botong Shang. 2020. The SEC's Deterrence Effect on Corporate Fraud. SSRN Electronic Journal 56. .
 [Crossref]
- 134. Matthias Kasper, James Alm. 2020. Audits, Audit Effectiveness, and Post-audit Tax Compliance. SSRN Electronic Journal. [Crossref]
- Adam Soliman. 2020. Crime and (a Preference for) Punishment: The Effects of Drug Policy Reform on Policing Activity. SSRN Electronic Journal 55. [Crossref]
- 136. Ross Deuchar, Johanne Miller, James Densley. 2019. The Lived Experience of Stop and Search in Scotland: There Are Two Sides to Every Story. *Police Quarterly* 22:4, 416-451. [Crossref]

- 137. Aaron Chalfin, Shooshan Danagoulian, Monica Deza. 2019. More sneezing, less crime? Health shocks and the market for offenses. *Journal of Health Economics* 68, 102230. [Crossref]
- 138. Joel Slemrod. 2019. Tax Compliance and Enforcement. *Journal of Economic Literature* 57:4, 904-954. [Abstract] [View PDF article] [PDF with links]
- 139. Laura Jaitman. 2019. Frontiers in the economics of crime: lessons for Latin America and the Caribbean. Latin American Economic Review 28:1. [Crossref]
- 140. Salvatore Capasso, Rajeev K. Goel, James W. Saunoris. 2019. Is it the gums, teeth or the bite? Effectiveness of dimensions of enforcement in curbing corruption. *Economics of Governance* 20:4, 329-369. [Crossref]
- 141. Greg Pogarsky, Shaina Herman. 2019. Nudging and the choice architecture of offending decisions. Criminology & Public Policy 18:4, 823-839. [Crossref]
- 142. Rosario Crinò, Giovanni Immordino, Salvatore Piccolo. 2019. Marginal deterrence at work. *Journal of Economic Behavior & Organization* 166, 586-612. [Crossref]
- 143. Jessica Di Salvatore. 2019. Peacekeepers against Criminal Violence—Unintended Effects of Peacekeeping Operations?. *American Journal of Political Science* 63:4, 840-858. [Crossref]
- 144. Griffin Edwards, Joshua J. Robinson. 2019. You gotta fight for your right? Publicly assigned but privately enforced property rights. *International Review of Law and Economics* 59, 31-39. [Crossref]
- 145. Ian Hoffman, Evan Mast. 2019. Heterogeneity in the effect of federal spending on local crime: Evidence from causal forests. *Regional Science and Urban Economics* **78**, 103463. [Crossref]
- 146. Viviane Sanfelice. 2019. Are safe routes effective? Assessing the effects of Chicago's Safe Passage program on local crimes. *Journal of Economic Behavior & Organization* 164, 357-373. [Crossref]
- 147. Justin McCrary, Deepak Premkumar. Why We Need Police 65-84. [Crossref]
- 148. Alejandro Cid. 2019. Saturation Policing and Robberies: Quasi-Experimental Evidence about the Effect of Sudden and Quick Operations. *Justice Evaluation Journal* 2:2, 164-180. [Crossref]
- 149. John J. Donohue, Abhay Aneja, Kyle D. Weber. 2019. Right-to-Carry Laws and Violent Crime: A Comprehensive Assessment Using Panel Data and a State-Level Synthetic Control Analysis. *Journal* of Empirical Legal Studies 16:2, 198-247. [Crossref]
- 150. Grichawat Lowatcharin, Judith I. Stallmann. 2019. The differential effects of decentralization on police intensity: A cross-national comparison. *The Social Science Journal* 56:2, 196-207. [Crossref]
- Michael Tonry. 2019. Fifty Years of American Sentencing Reform: Nine Lessons. *Crime and Justice* 48, 1-34. [Crossref]
- 152. Michael Tonry. 2019. Predictions of Dangerousness in Sentencing: Déjà Vu All Over Again. Crime and Justice 48, 439-482. [Crossref]
- 153. A. Mitchell Polinsky, Paul N. Riskind. 2019. Deterrence and the Optimal Use of Prison, Parole, and Probation. *The Journal of Law and Economics* 62:2, 347-371. [Crossref]
- 154. Emily K Weisburst. 2019. Safety in Police Numbers: Evidence of Police Effectiveness from Federal COPS Grant Applications. *American Law and Economics Review* 21:1, 81-109. [Crossref]
- 155. José R. Bucheli, Matías Fontenla, Benjamin James Waddell. 2019. Return migration and violence. World Development 116, 113-124. [Crossref]
- 156. Steven Mello. 2019. More COPS, less crime. Journal of Public Economics 172, 174-200. [Crossref]

- 157. Brian Beach, John Lopresti. 2019. LOSING BY LESS? IMPORT COMPETITION, UNEMPLOYMENT INSURANCE GENEROSITY, AND CRIME. *Economic Inquiry* 57:2, 1163-1181. [Crossref]
- 158. Aaron Chalfin, Monica Deza. 2019. The intergenerational effects of education on delinquency. *Journal of Economic Behavior & Organization* **159**, 553-571. [Crossref]
- 159. Daniel McMillen, Ignacio Sarmiento-Barbieri, Ruchi Singh. 2019. Do more eyes on the street reduce Crime? Evidence from Chicago's safe passage program. *Journal of Urban Economics* 110, 1-25. [Crossref]
- 160. José M. Alonso, Rhys Andrews, Vanesa Jorda. 2019. Do neighbourhood renewal programs reduce crime rates? Evidence from England. *Journal of Urban Economics* 110, 51-69. [Crossref]
- 161. Jacob Kaplan, Aaron Chalfin. 2019. More cops, fewer prisoners?. Criminology & Public Policy 18:1, 171-200. [Crossref]
- 162. Carl Kitchens, Matthew Philip Makofske, Le Wang. 2019. "Crime" on the Field. Southern Economic Journal 85:3, 821-864. [Crossref]
- 163. Jacob Nussim. Avoidance 1-14. [Crossref]
- 164. Sofie Pilemalm. ICT-Enabled Citizen Co-production in Excluded Areas Using Volunteers in Emergency Response 87-101. [Crossref]
- 165. Robert Dur, Ben Vollaard. 2019. Salience of law enforcement: A field experiment. *Journal of Environmental Economics and Management* 93, 208-220. [Crossref]
- 166. Murat C. Mungan. 2019. Positive Sanctions versus Imprisonment. SSRN Electronic Journal . [Crossref]
- 167. Francesco D'Acunto, Michael Weber, Jin Xie. 2019. Punish One, Teach A Hundred: The Sobering Effect of Punishment on the Unpunished. *SSRN Electronic Journal*. [Crossref]
- 168. Federico Masera. 2019. Violent Crime and the Overmilitarization of US Policing. SSRN Electronic Journal . [Crossref]
- 169. Alex Raskolnikov. 2019. Deterrence Theory: Key Findings and Challenges. SSRN Electronic Journal . [Crossref]
- 170. Eugen Dimant, Gerben A. van Kleef, Shaul Shalvi. 2019. Requiem for a Nudge: Framing Effects in Nudging Honesty. SSRN Electronic Journal . [Crossref]
- 171. Marie Claire Villeval. 2019. Comportements (non) éthiques et stratégies morales. *Revue économique* **70**:6, 1021. [Crossref]
- 172. Sam Krumholz. 2019. Enforcing Compliance: The Case of Automatic License Suspensions. SSRN Electronic Journal. [Crossref]
- 173. Matteo Rizzolli. Type-I and Type-II Errors 2112-2120. [Crossref]
- 174. Jacob Nussim. Avoidance 102-115. [Crossref]
- 175. Caroline Palmer, David C. Phillips, James X. Sullivan. 2019. Does emergency financial assistance reduce crime?. *Journal of Public Economics* 169, 34-51. [Crossref]
- 176. A. Mitchell Polinsky, Steven Shavell. 2019. Deterrence and the Adjustment of Sentences During Imprisonment. SSRN Electronic Journal 76. . [Crossref]
- 177. Matteo Rizzolli, James Tremewan. 2018. Hard labor in the lab: Deterrence, non-monetary sanctions, and severe procedures. *Journal of Behavioral and Experimental Economics* **77**, 107-121. [Crossref]

- 178. Nadia Campaniello, Evelina Gavrilova. 2018. Uncovering the gender participation gap in crime. *European Economic Review* 109, 289-304. [Crossref]
- 179. Chao Fu, Kenneth I Wolpin. 2018. Structural Estimation of a Becker-Ehrlich Equilibrium Model of Crime: Allocating Police Across Cities to Reduce Crime. *The Review of Economic Studies* 85:4, 2097-2138. [Crossref]
- 180. Joseph J. Sabia, Thanh Tam Nguyen. 2018. The Effect of Medical Marijuana Laws on Labor Market Outcomes. *The Journal of Law and Economics* **61**:3, 361-396. [Crossref]
- 181. Brian Bell, Anna Bindler, Stephen Machin. 2018. Crime Scars: Recessions and the Making of Career Criminals. *The Review of Economics and Statistics* 100:3, 392-404. [Crossref]
- 182. Rodney J. Andrews, Monica Deza. 2018. Local natural resources and crime: Evidence from oil price fluctuations in Texas. *Journal of Economic Behavior & Organization* 151, 123-142. [Crossref]
- 183. Yi Che, Xun Xu, Yan Zhang. 2018. Chinese import competition, crime, and government transfers in US. *Journal of Comparative Economics* **46**:2, 544-567. [Crossref]
- 184. Randi Hjalmarsson, Matthew J. Lindquist. 2018. Labour economics and crime. Labour Economics 52, 147-148. [Crossref]
- 185. Eberhard Feess, Hannah Schildberg-Hörisch, Markus Schramm, Ansgar Wohlschlegel. 2018. The impact of fine size and uncertainty on punishment and deterrence: Theory and evidence from the laboratory. *Journal of Economic Behavior & Organization* 149, 58-73. [Crossref]
- 186. Jordi Blanes i Vidal, Tom Kirchmaier. 2018. The Effect of Police Response Time on Crime Clearance Rates. *The Review of Economic Studies* 85:2, 855-891. [Crossref]
- 187. Matteo Rizzolli. Type-I and Type-II Errors 1-9. [Crossref]
- 188. Claudio Ferraz, Frederico Finan. Fighting Political Corruption: Evidence from Brazil 253-284. [Crossref]
- 189. Julius Heß. Herr Kagame träumt von Singapur. Eine globale quantitative Analyse zum Verhältnis von Freiheit, Sicherheit und Staatlichkeit 11-50. [Crossref]
- 190. John Braithwaite. 2018. Minimally Sufficient Deterrence. Crime and Justice 47:1, 69-118. [Crossref]
- 191. Benjamin Hansen, Keaton Miller, Caroline Weber. 2018. The Grass is Greener on the Other Side: How Extensive is the Interstate Trafficking of Recreational Marijuana?. SSRN Electronic Journal . [Crossref]
- 192. Stephen G. Dimmock, William Christopher Gerken, Tyson D. Van Alfen. 2018. Real Estate Shocks and Financial Advisor Misconduct. *SSRN Electronic Journal*. [Crossref]
- 193. Jacob Kaplan, Aaron Chalfin. 2018. More Cops, Fewer Prisoners?. SSRN Electronic Journal . [Crossref]
- 194. Federico Curci, Federico Masera. 2018. Flight From Urban Blight: Lead Poisoning, Crime and Suburbanization. SSRN Electronic Journal. [Crossref]
- 195. Claudia M. Landeo, Kathryn E. Spier. 2018. Ordered Leniency: An Experimental Study of Law Enforcement with Self-Reporting. *SSRN Electronic Journal*. [Crossref]
- 196. Sebastian Blesse, André Diegmann. 2018. Police Reorganization and Crime: Evidence from Police Station Closures. SSRN Electronic Journal . [Crossref]
- 197. Ashna Arora. 2018. Juvenile Crime and Anticipated Punishment. SSRN Electronic Journal . [Crossref]

- 198. Allison Stashko. 2018. Do Police Maximize Arrests or Minimize Crime? Evidence from Racial Profiling in U.S. Cities. SSRN Electronic Journal. [Crossref]
- 199. Ignacio Sarmiento-Barbieri, Qi Ge, Rodrigo Schneider. 2018. Emotional Cues and Crime: Spatial and Temporal Evidence from Brazilian Soccer Games. *SSRN Electronic Journal* . [Crossref]
- 200. Johannes Buckenmaier, Eugen Dimant, Ann-Christin Posten, Ulrich Schmidt. 2018. Efficient Institutions and Effective Deterrence: On Timing and Uncertainty of Punishment. SSRN Electronic Journal. [Crossref]
- 201. Daniel S. Nagin, Cody W. Telep. 2017. Response to "Procedural Justice and Policing: A Rush to Judgment?". Annual Review of Law and Social Science 13:1, 55-58. [Crossref]
- 202. Liuchun Deng, Yufeng Sun. 2017. Criminal network formation and optimal detection policy: The role of cascade of detection. *Journal of Economic Behavior & Organization* 141, 43-63. [Crossref]
- 203. Vincenzo Bove, Evelina Gavrilova. 2017. Police Officer on the Frontline or a Soldier? The Effect of Police Militarization on Crime. *American Economic Journal: Economic Policy* 9:3, 1-18. [Abstract] [View PDF article] [PDF with links]
- 204. Santiago Gomez-Cardona, Daniel Mejia, Santiago Tobon. 2017. The Deterrent Effect of Public Surveillance Cameras on Crime. SSRN Electronic Journal. [Crossref]
- 205. A. Mitchell Polinsky. 2017. Deterrence and the Optimal Use of Prison, Parole, and Probation. SSRN Electronic Journal . [Crossref]
- 206. Shawn Kantor, Carl Kitchens, Steven Pawlowski. 2017. Civil Asset Forfeiture, Crime, and Police Incentives: Evidence from the Comprehensive Crime Control Act of 1984. SSRN Electronic Journal . [Crossref]
- 207. Christopher Blattman, Donald P. Green, Daniel Ortega, Santiago Tobon. 2017. Pushing Crime Around the Corner? Estimating Experimental Impacts of Large-Scale Security Interventions. SSRN Electronic Journal. [Crossref]
- 208. Christian Matthew Leister, Yves Zenou, Junjie Zhou. 2017. Coordination on Networks. SSRN Electronic Journal. [Crossref]
- 209. Radek Janhuba, Kristyna Cechova. 2017. Criminals on the Field: A Study of College Football. SSRN Electronic Journal . [Crossref]
- 210. Nicholas Lovett, Yuhan Xue. 2017. Do Greater Sanctions Deter Youth Crime? Evidence from a Regression Discontinuity Design. SSRN Electronic Journal . [Crossref]
- 211. Paul Liu, Marco Fabbri. 2016. More Eyes, (No Guns,) Less Crime: Estimating the Effects of Unarmed Private Patrols on Crime Using a Bayesian Structural Time-Series Model. SSRN Electronic Journal . [Crossref]
- 212. Roee Sarel. 2016. Judicial Errors, Crime Deterrence and Appeals: Evidence from U.S. Federal Courts. SSRN Electronic Journal . [Crossref]
- 213. Gregory J. DeAngelo, Brad R. Humphreys, Imke Reimers. 2016. Are Public and Private Enforcement Complements or Substitutes? Evidence from High Frequency NHL Data. SSRN Electronic Journal . [Crossref]
- 214. Steven Mello. 2016. Police and Crime: Evidence from Cops 2.0. SSRN Electronic Journal . [Crossref]
- 215. Emily Weisburst. 2016. Safety in Police Numbers: Evidence of Police Effectiveness and Foresight from Federal COPS Grant Applications. SSRN Electronic Journal. [Crossref]

- 216. Federico Masera. 2016. Bringing War Home: Violent Crime, Police Killings and the Overmilitarization of the US Police. *SSRN Electronic Journal*. [Crossref]
- 217. Robert Dur, Ben Vollaard. 2016. Salience of Law Enforcement: A Field Experiment. SSRN Electronic Journal . [Crossref]
- 218. Sandra V. Rozo, Steven P. Raphael, Therese Anders. 2016. Deportation, Crime, and Victimization. SSRN Electronic Journal. [Crossref]
- 219. Jennifer L. Doleac, Rasmus Landerss, Anne Sofie Tegner Anker. 2016. The Effects of DNA Databases on the Deterrence and Detection of Offenders. *SSRN Electronic Journal*. [Crossref]
- 220. Jennifer L. Doleac. 2015. The Effects of DNA Databases on Crime. SSRN Electronic Journal . [Crossref]
- 221. Lones Smith, Jorge VVsquez. 2015. Crime and Vigilance. SSRN Electronic Journal . [Crossref]
- 222. Francisco Costa, Jooo S de Faria, Felipe Iachan. 2015. Homicides and the Age of Criminal Responsibility in Brazil: A Regression Discontinuity Approach. SSRN Electronic Journal. [Crossref]
- 223. Vincenzo Bove, Evelina Gavrilova. 2015. Policeman on the Frontline or a Soldier? The Effect of Police Militarization on Crime. SSRN Electronic Journal. [Crossref]
- 224. Javier Donna, Joss-Antonio Esppn-SSnchez. 2014. Let the Punishment Fit the Criminal. SSRN Electronic Journal 23. . [Crossref]
- 225. Alain Cohn, Michel Andrr Marrchal, Thomas Noll. 2013. Bad Boys: The Effect of Criminal Identity on Dishonesty. *SSRN Electronic Journal*. [Crossref]